# PENSIONS POLICY INSTITUTE

Retirement income and assets: the implications for retirement income of Government policies to extend working lives



A Discussion Paper by Daniela Silcock, Daniel Redwood and Chris Curry

Published by the Pensions Policy Institute © April 2012 ISBN 978-1-906284-22-0 www.pensionspolicyinstitute.org.uk

### Retirement income and assets: the implications for retirement income of Government policies to extend working lives

Executive Summary		1
Int	roduction	8
1.	What are the current patterns of work and retirement in the UK?	10
2.	What are the implications of the Government's extending working lives policy agenda? What can we learn from international experience of similar policies?	29
3.	How much longer might individuals need to work and save in order to meet target levels of retirement income?	49
4.	How might different patterns of work and retirement impact the pension income of individuals?	65
Appendix – Modelling analysis		
Acknowledgements and contact details		85
Ref	ferences	87

#### **Executive Summary**

#### Introduction

In the UK over the last three decades, life expectancy has increased substantially. In 1981 a 65 year old man could have expected to live for another 14 years on average, to age 79. However in 2011 a 65 year old man can reasonably expect to live for another 21 years, to age 86 on average.

These substantial improvements in life expectancy reflect a complex range of changes in lifestyle, diet, healthcare and patterns of work and economic activity and could be positive for individuals. However, increased life expectancy also poses significant challenges to individuals, to employers and to the Government. Individuals who live longer may have increasingly long retirements to save for and support, employers who sponsor Defined Benefit schemes face increases in the costs of providing these pensions and the Government may face increased pressure from funding the state pension and benefits for pensioners.

This research examines current patterns of work and retirement in the UK among people over age 50 and considers how these might change in the future. The Government has introduced a range of policies and programmes aimed at extending working lives such as policies aimed at moving people off benefits and into paid work, changes to the age at which people can receive state pension and benefit income, and the removal of legal barriers to working longer. The research looks at international experience of similar policy changes and considers what lessons the UK might learn from them. Finally, the research considers what patterns of work and retirement might mean for income adequacy in retirement.

### The proportion of people over age 50 in work has risen over the last few decades in the UK

Employment rates have been increasing for people aged 50 and above in the UK. For example, in 1993, around 64% of men aged 50 to 64 were in work in the UK, by 2011 this had increased to 70%. The proportion of men working beyond the current male State Pension Age of age 65 has also increased. In 1993 around 8% of men over age 65 were in work, by 2011 this had increased to 11%.

The increase in employment rates at older ages has been even more substantial for women. The State Pension Age for women was age 60 until 2010, but is now increasing towards the male State Pension Age of 65 under legislation intended to equalise the male and female State Pension Age at age 65 by 2018.

In 1993, around 57% of women aged 50 to 59 in the UK were in work, by 2011 this had increased to around 72%. The proportion of women

working beyond the women's State Pension Age of age 60 has also increased substantially. In 1993, 23% of women aged between 60 and 64 were in work, by 2011 this had increased to 34%.

The average age of exit from the labour market has been increasing for both men and women. In 1984, the average age of exit from the labour market for men in the UK was age 64, by 2011 this had increased to around age 65 for men. For women, the average age of exit from the labour market was around age 61 in 1984 and increased to age 63 by 2011.

These increases in employment beyond age 50 for both men and women mask considerable differences in employment rates at older ages by occupation, skill level and wealth. Working at age 50 and above is more likely for men, for those with higher wealth, for those with higher education and for the self-employed.

#### Why do people leave work before reaching State Pension Age?

While the proportions of people working at older ages has increased, many people are compelled to leave work before State Pension Age due to circumstances beyond their control, such as health problems or the need to provide care for a family member.

Health problems are one of the main, non-voluntary reasons for people to leave work before State Pension Age. By the time men and women are aged between 60 and 64, around 30% of them have a disability that limits their ability to work. Age, gender, ethnicity, occupation and location affect both the likelihood of having a work limiting disability and the likelihood of continuing to work while disabled. Work limiting disability is more common among older people in lower wealth quintiles, those with lower levels of education and manual workers who are more likely than those in other occupations to leave work due to health problems.

Providing care can also affect people's ability to work at older ages; in 2008/9, 10% of women and 1% of men aged 50 and older who were not in employment were providing care.

Equally some people choose to leave work voluntarily before reaching their State Pension Age. Men are more likely to retire voluntarily before reaching their State Pension Age than women. In 2008/9, around 29% of men reported that they were retiring voluntarily in the five years before State Pension Age, compared to around 8% of women.

People in the highest wealth quintile are more than twice as likely to retire voluntarily before reaching their State Pension Age, than people in the middle wealth quintile. Those with a Defined Benefit pension are almost twice as likely to retire voluntarily before their State Pension Age than those with no private pension income.

In order for older workers to be able to engage in employment there needs to be an appetite from employers to recruit and retain older workers and employers need to be able to provide appropriate support to those older people who need it.

### Trends and changes within private pensions could affect future work and retirement patterns

Within the private sector, many of the employers who offer Defined Benefit (DB) pensions have closed their schemes to new members, and increasingly to existing members too, as the costs and risks of providing this type of pension have increased. Increasingly employers are offering current employees Defined Contribution (DC) pension schemes. Autoenrolment into private pensions, which begins in 2012, will also lead to a substantial increase in people saving in DC pensions.

Members of DC schemes may have more incentive to work longer than members of DB schemes, as working longer can lead to increases in income in retirement from a DC scheme and because levels of contribution into DC schemes have tended to be lower than to DB schemes. This could mean that some people may have to work for longer than expected in order to ensure sufficient income during working life and save for longer into a pension to ensure they have adequate incomes in retirement.

Some policies aimed at extending working lives could increase income for those who can work longer, but may disadvantage those who cannot The Government is keen to encourage people to work longer in order to combat the potential financial problems both for the individual and for the state posed by a declining birth rate and increases in longevity. The Government has a range of policies and programmes aimed at extending working lives, including:

- policies aimed at moving people off benefits and into paid work;
- changes to the age at which people can receive state pension and benefits income;
- the removal of barriers to working longer, including the removal of legal barriers and efforts to tackle age discrimination in the workplace.

While these policies will encourage and enable some people to work longer, and to take their state and private pension later, there may be negative financial implications for those who cannot work longer because of health problems, or other factors such as caring responsibilities. For those who are unable to work longer, rises to the State Pension Age and changes to the benefit system may result in lower incomes in retirement.

Changes to the benefit system may affect older people in particular. Around a million people who are over age 50 are currently receiving Incapacity Benefits. Over the next few years these people will be phased on to Employment and Support Allowance (ESA) which has a stricter assessment regime. Being found Fit for Work is no guarantee of actually being able to find work. Of those aged 55 and over who were found Fit for Work under the ESA assessment (or had their claim dropped) only around a quarter were in work a year to 16 months later.

A number of other countries have undertaken policy reforms to encourage extending working lives similar to the policies being implemented in the UK. For example:

- Denmark is formally linking increases in State Pension Age to increases in Life Expectancy from 2015. Under current proposals State Pension Age in Denmark is expected to increase to 72 by 2055.
- The USA removed mandatory retirement during the 1970s and 1980s. This has contributed to increases in employment above age 65, which has risen from 11% of people aged 65 and over in 1985 to 17% in 2010.
- New Zealand introduced policies aimed at extending working lives from the early 1990s onwards. New Zealand increased its State Pension Age, removed mandatory retirement on the basis of age and introduced a positive ageing strategy. Although other factors have also played a significant role in increasing participation at older ages including a skills shortage, the increase in the employment rate among older workers increased dramatically. In 1990, 26% of people aged 60 to 64 were in employment in New Zealand. By 2000, this had risen to 46% and by 2010, 70% of people aged 60 to 64 in New Zealand were in employment.

Although the environment and context in the countries where these policies have been implemented differs from that in the UK, there are still some policy conclusions for the UK:

- It is important to have safeguards built into policies designed to encourage working at older ages to help those individuals who cannot work longer.
- Other factors, such as economic changes and changes in pension design, are also likely to be important in determining participation rates at older ages.
- Increases in life expectancy, even if accompanied by increases in working at older ages, are likely to need to be accompanied by increased saving if individuals are to have an adequate income in retirement.

### What could future patterns of work and retirement mean for retirement income adequacy?

This research uses two measures of income adequacy in order to test whether pensioners might have enough income in order to meet their needs in retirement:

- The **Minimum Income Standard**, which calculates how much income pensioners require to meet a *minimum acceptable standard of living* (as defined by pensioners in focus groups) is just under £11,000pa (£211pw) for a single pensioner Before Housing Costs (BHC), and around £15,700pa (£303pw) for a couple in 2011. For people who have lived on medium to high incomes during working life, the minimum acceptable standard of living may not seem adequate as it would generally constitute a drop in living standards.
- Working life replacement rates which calculate how much income an individual pensioner might need in order to achieve a similar standard of living to the one they had in working life. Replacement rates are generally in the range of 50% to 80% of people's gross working life income. A median-earning man with a weekly income at the point of retirement of around £500pw might need a gross weekly retirement income of around £17,400pa (£335pw) to meet a 67% replacement rate of working life income (and recreate working life living standards). In this analysis, the lowest replacement rates are aligned with the Minimum Income Standard, so that no one has a target income below the Minimum Income Standard.

In order to examine how working longer can affect the adequacy of retirement income, the PPI has modelled the retirement income that individuals aged between 50 and State Pension Age in 2011 might expect to achieve if they continue to work and save at current levels, up until or beyond their State Pension Age.

The analysis assumes that the individuals remain in work, and continue to earn and save at their existing levels until they hit the target level of retirement income. In reality of course, some people may not be able to continue to work and save longer, so the analysis is illustrative of the extent of longer working that may be needed to achieve adequate retirement incomes, rather than a projection of what will actually happen.

The analysis is based on the English Longitudinal Study of Ageing data set of people still in work aged between 50 and State Pension Age in England. Where individuals are not currently saving in a pension it is assumed that, if eligible, they are automatically enrolled into pension saving at minimum required levels – (which are phased in between 2012 and 2019 to reach 8% minimum total contributions on band earnings by 2019).

The following results have been rounded to the nearest 5%. Totals may not sum to 100% because of this rounding.

# How many people aged between 50 and State Pension Age in 2011 might be able to meet a Minimum Income Standard of retirement income?

Meeting the Minimum Income Standard will be easier for people than meeting a target replacement rate of working life income using only state pension and benefits and private pension income:

- The vast majority, around 85%, of those aged between 50 and State Pension Age in 2011 and still in work might have sufficient state and private pension income to meet the Minimum Income Standard by their State Pension Age assuming that everyone continues to work and save until their State Pension Age and that those who are entitled to means-tested benefits claim them.
- Means-tested benefits can play an important role in helping those on low incomes during working life achieve minimum acceptable standards of living in retirement. Some of those who don't meet the Minimum Income Standard are not eligible for means-tested benefits as their savings are too high.
- Around 10% of those aged between 50 and State Pension Age in 2011 and still in work will only be able to meet the Minimum Income Standard if they continue to work and save for a further one to five years after State Pension Age.
- Around 5% of those aged between 50 and State Pension Age in 2011 and still in work might need to work for six years or more after their State Pension Age in order to meet the Minimum Income Standard.

### How many people aged between age 50 and State Pension Age in 2011 might be able to meet a target replacement rate of retirement income?

Fewer people will be able to meet a target replacement rate of working life income that would allow them to replicate working life living standards by their State Pension Age, using only state pension and benefit income and private pension income:

- Around 40% of those aged between 50 and State Pension Age in 2011 and still in work might have sufficient state and private pension income to meet a target working life replacement rate by their State Pension Age.
- Around 10% of those aged between 50 and State Pension Age in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further one to five years after State Pension Age.
- Around 5% of those aged between 50 and State Pension Age in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further six to ten years after State Pension Age.
- Around 45% of those aged between 50 and State Pension Age in 2011 and still in work, might need to work and save for 11 or more years after their State Pension Age in order to meet a target replacement rate of working life income.

People with certain characteristics may be more or less likely to be able to meet a target replacement rate by State Pension Age from state and private pension income:

- Those in the lowest income quartile may be more able to meet a target replacement rate of working life income by State Pension Age, as state pension and benefit levels can be close to or above their target replacement rate levels.
- Single women and people in couples are more likely to meet a target replacement rate than single men. For single women this is because they are more likely to have low incomes than single men, and couples can benefit from pooling incomes.
- People with Defined Benefit pension savings might find it easier to meet a target replacement rate in retirement, as historically Defined Benefit pensions have been more generous than Defined Contribution pensions.
- Contributing more to a Defined Contribution pension could mean that people do not need to work as long to meet a replacement rate of working life income which would allow them to replicate working life living standards.

Further modelling analysis explored the impact of working longer and not working longer on 3 hypothetical individuals who have different income levels and saving histories during their working life. The modelling analysis illustrated that:

- Leaving work before State Pension Age can result in lower income both before and after State Pension Age.
- Shopping around and purchasing an enhanced annuity (for example, an annuity offered to individuals with a medical condition that pays a higher annual pension in recognition that their life expectancy is shorter than average) could increase income in retirement for people with health problems.
- Disability benefits can play an important role in meeting income needs for those who have to leave work early due to health problems.
- Remaining in work until State Pension Age can help to maintain living standards up until retirement.
- A high earner, contributing at average levels of salary into a Defined Contribution pension may need to work beyond State Pension Age in order to meet and sustain his target income during retirement.
- Working after State Pension Age can increase net income as a result of tax treatment.

The analysis explored how changes in behaviour in response to a rise in State Pension Age can affect income both before and after retirement:

- As expected, a rise in State Pension Age could reduce income for those who cannot work longer.
- But a rise in State Pension Age could result in higher income both before and after retirement for those who can work longer.

#### Introduction

In the UK over the last three decades, life expectancy has increased substantially. In 1981 a 65 year old man could have expected to live for another 14 years on average, to age 79.<sup>1</sup> However in 2011 a 65 year old man can reasonably expect to live for another 21 years, to age 86 on average.<sup>2</sup>

These substantial improvements in life expectancy reflect a complex range of changes in lifestyle, diet, healthcare and patterns of work and economic activity. Improvements in life expectancy could be positive for individuals. However, increased life expectancy also poses significant challenges to individuals, to employers and to the Government. Individuals who live longer may have increasingly long retirements to save for and support, employers who sponsor Defined Benefit schemes face increases in the costs of providing these pensions and the Government may face increased pressure from funding the state pension and benefits for pensioners.

The Government has introduced several policies aimed at extending working lives. Changes to the State Pension Age in particular are likely to impact on future patterns of work, saving and retirement. Other Government policies (such as removing the Default Retirement Age) and trends within private pensions (such as changes to public sector pension schemes and the private sector shift from Defined Benefit to Defined Contribution) are also likely to affect the length of people's working lives and their retirement income.

This report draws together the latest information on current patterns of work and retirement in the UK. It also describes the Government's policy agenda to extend working lives and looks at the lessons the UK can learn from international experience. Finally, using the PPI's modelling capability, this paper considers how much longer individuals may need to work beyond State Pension Age if they wish to meet target levels of retirement income solely through state and private pension income.

Chapter one describes current patterns of work and retirement, and investigates the different social, political, economic and personal factors which impact the decisions people make and the options people have regarding work and retirement.

Chapter two describes and examines the potential implications of Government policies which will affect future work and retirement patterns for older people and looks at the lessons that the UK can learn from other countries that have implemented similar policies.

<sup>&</sup>lt;sup>1</sup> 2010 based cohort expectation of life, 1981 to 2060, principal projection, United Kingdom, ONS <sup>2</sup> 2010 based cohort expectation of life, 1981 to 2060, principal projection, United Kingdom, ONS

Chapter three analyses how working and retirement patterns could impact on future levels of income adequacy for people in retirement.

Chapter four analyses how different patterns of work and retirement could affect the income of individuals who have different income levels.

#### <u>Chapter one: what are the current patterns of</u> work and retirement in the UK?

This chapter describes current patterns of work and retirement, and investigates the different social, political, economic and personal factors which impact the decisions people make and the options people have regarding work and retirement.

### For the purposes of this research the term 'older people' refers to people over age 50

This chapter investigates work and retirement trends amongst all people over the age of 50, 'older people', and also separately investigates trends among those over State Pension Age (SPA) which is currently rising from age 60 to 65 for women and is age 65 for men. These are both useful groups to examine. Working trends among people over the age of 50 can tell us about the impact of economic and policy changes on general working at older ages, for example, by exploring rises in the average age of labour-market exit. It is also useful to explore trends among those over SPA as SPA is a natural choice as a retirement age for many people.<sup>3</sup>

Figures in this chapter have been drawn from two main sources. High level information on employment has been estimated using the Labour Force Survey (LFS), which covers the UK. More detailed analysis of movements within the Labour Market and analysis of retirement is generally based on the English Longitudinal Study of Ageing (ELSA), which covers only England.

The next section of this chapter investigates how retirement ages and the proportion of older people in work have changed over the last two decades.

### The proportion of older people in work has risen over the last few decades

Employment rates have been increasing for people over the age of 50 since the mid-1990s. While some people may work until or beyond SPA because they enjoy their work, other people work at older ages because they feel that they cannot afford to retire for financial reasons.<sup>4</sup>

<sup>&</sup>lt;sup>3</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>4</sup> AIFA (2010)

#### Employment rates have been growing for men at older ages

There has been an increase in working at older ages among men:

- In 1993, around 64% of men in the UK aged 50 to 64 were in work. In 2011, this had increased to around 70% of men aged 50 to 64 in work.<sup>5</sup>
- In 1993, around 8% of men in the UK over the current male SPA (of age 65) were in work. In 2011, this had increased to around 11% of men over the age of 65 in work.<sup>6</sup>

Chart 17



The increase in employment is especially high for men in age groups just before and just after Men's State Pension Age (Chart 1).

- In 1993, 44% of men in the UK aged between 60 and 64, in the run up to men's SPA, were in employment which increased to 54% by 2011, an increase of 10 percentage points.
- In 1993, 12% of men in the UK aged between 65 and 69, the 5 years just after SPA, were in employment which increased to 22% by 2011, an increase of 10 percentage points.

However, even after these increases the number of men working at older ages is still relatively low. In 2011 there were 330,000 men aged between 65 and 69, 115,000 men aged between 70 and 74, and 60,000 men aged 75 or older in work in the UK.<sup>8</sup>

<sup>&</sup>lt;sup>5</sup> PPI analysis of LFS data

<sup>&</sup>lt;sup>6</sup> PPI analysis of LFS data

<sup>&</sup>lt;sup>7</sup> PPI analysis of LFS data – includes all full-time and part-time work

<sup>&</sup>lt;sup>8</sup> PPI analysis of LFS data - rounded to the nearest 5,000.

**Levels of working at older ages have increased significantly for women** Working at older ages has increased significantly over the last few decades for both men and women, but women have seen a faster increase:

- In 1993, around 57% of women in the UK aged between 50 and 59, the ten years below women's SPA of age 60, were in work. In 2011, around 72% of women in the UK aged between 50 and 59 in the UK were in work.<sup>9</sup>
- In 1993, around 8% of women in the UK over women's SPA of 60 were in work. In 2011, around 13% of women in the UK over age 60 were in work.<sup>10</sup>

#### Chart 211



There have been substantial increases in labour market participation for women over age 60 (Chart 2). The age groups that have seen the greatest increase in participation are women in the ten year run up to women's State Pension Age and the 4 years after women's State Pension Age.

- In 1993, 57% of women in the UK aged between 50 and 59, in the ten year run-up to SPA, were in employment which increased to 72% by 2011, an increase of 15 percentage points.
- In 1993, 23% of women in the UK aged between 60 and 64, in the 5 years after SPA, were in employment which increased to 34% by 2011, an increase of 11 percentage points.

<sup>9</sup> PPI analysis of LFS data

<sup>&</sup>lt;sup>10</sup> PPI analysis of LFS data

<sup>&</sup>lt;sup>11</sup> PPI analysis of LFS data - includes all full-time and part-time work

Again, despite the increase in employment rates the number of women in employment at older ages is still low. In 2011 there were 230,000 women aged between 65 and 69, 65,000 women aged between 70 and 74, and 10,000 women aged 75 or older in work in the UK.<sup>12</sup>

#### People are leaving work at older ages than they used to

The average age of leaving the labour market has increased in the UK for women from around age 61 in 1984 to around age 63 in 2011 and for men from around age 64 in 1984 to around age 65 in 2011 (Chart 3).<sup>13</sup>



### Working at age 50 and above is more likely for those with higher wealth and education

Older people are not a homogenous group and their experiences of, and likelihood of being engaged with, the labour market at older ages varies according to their characteristics. Working at age 50 and above is more likely for:

- Men over age 50 than women over age 50.
- Those with higher wealth with 24% of the richest quintile of people over age 50 in full-time work in 2008/9 and 14% of those in the poorest.<sup>15</sup>

<sup>&</sup>lt;sup>12</sup> PPI analysis of LFS data – rounded to the nearest 5,000.

<sup>&</sup>lt;sup>13</sup> ONS (2012) figure 4.10

<sup>&</sup>lt;sup>14</sup> ONS (2012) figure 4.10

<sup>&</sup>lt;sup>15</sup> Crawford, Tetlow (2010) p. 62, Table 2A3. rounded to nearest percent

- Those with higher education with 35% of those with the highest education in full-time work in 2008/9 and 16% of those with the lowest education.<sup>16</sup>
- The self-employed and people working for small employers (Chart 4)

   who are more likely to work until they are physically unable to do so
   than the people in other groups. However, the self-employed are a
   very diverse group, ranging from highly skilled professionals to
   people with small or contract based businesses<sup>17</sup> who might need to
   continue working to provide themselves and their families with an
   income.<sup>18</sup>
- Those in higher skilled occupations (Chart 4), though there is little difference between high and low skilled occupations in the very oldest age groups (aged 75 and older). Some individuals in the low skilled category at these ages may have classified themselves as higher skilled workers earlier in their life.

#### Chart 419



Chart 4 is based on self-reported occupation type, and excludes individuals who report themselves as retired, and so only include those individuals with some attachment to the labour market. The figures in Chart 4 are not therefore directly comparable with the employment rates

<sup>18</sup> Phillipson, Smith (2005)

<sup>&</sup>lt;sup>16</sup> Crawford, Tetlow (2010) p. 61, Table 2A.2. rounded to nearest percent. **Low education =** those who left school at or before UK compulsory school leaving (CSL) age for their cohort. **Mid education =** those who left school after CSL age but before age 19. **High education =** those who left school at or after age 19.

<sup>&</sup>lt;sup>17</sup> For example, building/decorating services or running a small shop

<sup>&</sup>lt;sup>19</sup> PPI analysis of Labour Force Survey data, July 2011

shown in Charts 1 and 2, but do give an indication of the differences between different occupational groups.

#### Working beyond State Pension Age is more likely for women, people with higher education, those without a long standing illness, and people whose partners are still working

Working beyond State Pension Age (SPA) is also more likely for people with certain characteristics:

- Working beyond SPA is more likely for people with higher levels of education.<sup>20</sup>
- Those *without* a long standing illness are more likely to work beyond SPA.<sup>21</sup>
- Those with partners still working are more likely to work after SPA than those whose partners have retired.<sup>22</sup>
- Though working at ages over 50 is more likely for men, working beyond SPA is more likely for women, however this could be because they have a lower SPA than men (though this will no longer be the case after 2018).<sup>23</sup> Although women who have never married appear to have similar pension entitlements to men who never married,<sup>24</sup> women who have been married are less likely to have paid into a pension,<sup>25</sup> and this may contribute to decisions around working if, for example, women feel they cannot afford to retire.<sup>26</sup>

#### People can transition from work to retirement by flexible working

There are many ways to phase the transition from full-time work into retirement through the use of flexible working. Flexible working can take many forms (Box 1).

<sup>&</sup>lt;sup>20</sup> Crawford, Tetlow (2010), 40% more likely than those with low education

<sup>&</sup>lt;sup>21</sup> Crawford, Tetlow (2010), Those who have a partner with a long standing illness are also more likely to work at older ages

<sup>&</sup>lt;sup>22</sup> Crawford, Tetlow (2010)

<sup>23</sup> Banks, Tetlow (2008)

<sup>&</sup>lt;sup>24</sup> DWP (2006)

<sup>&</sup>lt;sup>25</sup> Because, for example, of broken work history, low earnings or an expectation of depending on their husbands pension.

<sup>&</sup>lt;sup>26</sup> Banks, Tetlow (2008)

#### Box 1: Flexible working

Flexible working can take many forms, within three broad types of flexibility.<sup>27</sup>

Flexible working hours, such as:

- Part-time working, working shorter days, or fewer days in a week.
- Flexible hours, where people can choose which hours they work.
- Compressing hours, by allowing people to work long shifts in return for extra days off.
- Working only the hours that are necessary to do the job.
- Becoming self-employed (and setting one's own hours).
- Sharing a job between two workers.

#### Flexible work space, such as

• Working from home for some or all of the time.

Flexible work, such as:

• A shift to work which is less physically demanding or stressful.

#### Flexible working is increasing among workers aged 50 and older

There has been an increase in the proportion of workers aged 50 and older who are working flexibly. Using one definition of flexible working, the proportion of people aged 50 or older in the UK in employment who were working flexibly increased from around 30% in 2005 to 38% by 2010.<sup>28</sup>

This estimate does not explicitly include one of the most commonly considered forms<sup>29</sup> of flexible working, part-time working. This is because not all part-time working is necessarily flexible working. Part-time work might be considered flexible if the hours are as a result of agreement between the employer and employee, but might not be flexible if an employee is working part-time only because no full-time work is available.

However, despite the problems in identifying whether part-time workers are working flexibly or working shorter hours than they would prefer due to the lack of opportunity, analysis of the trends in part-time work can help identify potential trends in flexible working.

The next section of this chapter therefore looks at current trends in parttime working among older people and investigates the potential for developments in part-time working in the future.

<sup>&</sup>lt;sup>27</sup> Hedges, Sykes (DWP) (2009), and Age UK (2012)

<sup>&</sup>lt;sup>28</sup> Age UK estimates based on the LFS where flexible working is defined as one of: Flexi-time, annualised hours contract, term time working, job sharing, nine-day fortnight, 4.5 day week, zero hours contract, on-call working or home working. Part-time work is not explicitly included, although part-time workers are treated as working flexibly if they use one of the other options listed.
<sup>29</sup> Alongside self-employment Phillipson, Smith (2005)

<sup>16</sup> 

#### The use of part-time working is increasing at older ages

The proportion of older people in part-time work is increasing, with a greater proportional increase in part-time working among men than women (Table 1).

Table 1: The proportion of men and women in England over age 50 in part-time work by year<sup>30</sup>

	2002/3	2008/9
Men	8%	10%
Women	18%	19%

Between 2002/3 and 2008/9 around 10% of people in England over age 50 moved from full-time work to part-time work (Table 2), and around 20% of these people then moved into retirement from part-time work.<sup>31</sup> The following table shows the transitions men and women in employment made between 2002/3 and 2008/9.

Table 2:	The employment	transitions of	of men	and wo	omen in i	England
between	age 50 and State P	'ension Age b	oetween	2002/3	and 2008/	<b>9</b> 32

	0	, , ,
Type of work, non-	Men	Women
work or transition		
during 2002/3 –		
2008/9		
Always full-time	30%	11%
Always part-time	3%	16%
Always inactive	26%	26%
Full-time to part-time	7%	7%
Full-time to part-time	2%	2%
to inactive		
Full-time to inactive	15%	7%
Part-time to inactive	4%	16%
Other	13%	15%

#### Women are more likely to work part-time than men

The likelihood of working part-time varies between different groups:

- Women are more likely to work part-time than men. In 2011, 30% of women in the UK and 5% of men aged between 50 and 54 were in part-time work (Chart 5).
- The self-employed are more likely to work part-time than employees,<sup>33</sup> though it may also be the case that people wishing to work part-time may become self-employed if there is a lack of opportunity for employed part-time work.

<sup>&</sup>lt;sup>30</sup> Crawford, Tetlow (2010) Table 2a.1

<sup>&</sup>lt;sup>31</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>32</sup> Crawford, Tetlow (2010) p. 72, Table 2A.12. rounded to nearest percent. Other includes all other

work patterns not already listed.

<sup>&</sup>lt;sup>33</sup> Banks, Tetlow (2008)

- Older people are more likely to work part-time than younger people; beyond age 65 part-time work is more common for both men and women than full-time work.<sup>34</sup>
- People in good health and with higher levels of education are more likely to phase into retirement by moving from full-time work to parttime work before retiring outright.<sup>35</sup>
- People who are entitled to a Defined Benefit pension, on the other hand, are more likely to retire outright from full-time work rather than to phase through part-time work first.<sup>36</sup>



#### Flexible working is not an option for everyone

A number of organisations<sup>38</sup> actively promote flexible working and the recruitment of older workers. In 2003, around 46% of employees reported having access to flexible working.<sup>39</sup> However flexible working is not available to all employees across all sectors. Employers in some sectors may find it harder than others to offer flexible working. Larger employers may find it easier to accommodate the need for flexible working through reducing hours or changing people's roles or job descriptions.

<sup>&</sup>lt;sup>34</sup> Banks, Tetlow (2008)

<sup>&</sup>lt;sup>35</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>36</sup> Crawford, Tetlow (2010)

 $<sup>^{\</sup>rm 37}$  PPI analysis of Labour Force Survey data, July 2011

<sup>&</sup>lt;sup>38</sup> For example, ASDA and B&Q

<sup>&</sup>lt;sup>39</sup> Phillipson, Smith (2005)

### The way people take their pension can impact access to flexible working

Pensions can play a role in access to flexible working. Since 2006 the requirement for people to leave the employer who manages their pension scheme before receiving a pension from that scheme has been removed, meaning that more flexible ways of working and receiving income could be available than were previously. For example, some people can now work part-time for the same employer, while taking their occupational pension to supplement their income (although not all employers offer this option). People with private Defined Contribution pensions can generally convert some or all of their fund into an income at any point from age 55 and use it to supplement earnings.

#### Why are some people working longer?

#### Many people work longer because they need income

For many people, working at older ages or beyond State Pension Age (SPA) will be a necessity rather than a choice. Many people work longer because they have financial commitments such as mortgages, children at university, non-working partners or other dependents.<sup>40</sup> Those with a mortgage are twice as likely as those who own their homes outright to work after SPA.<sup>41</sup> Many people also work longer because they feel they simply cannot afford to retire, or want to ensure that they have saved enough in their pension to provide themselves with an acceptable income in retirement.<sup>42</sup>

#### Attitudes to work have an impact on whether people work at older ages

As well as being associated with characteristics such as wealth and education level, the attitudes people have to work will be associated with their likelihood of working at older ages. Those who have a strong identification with work are more likely to continue working at older ages.<sup>43</sup> People have also cited the following wide-ranging reasons for staying in work at older ages:<sup>44</sup>

- maintaining social contact,
- keeping motivated and having something to do with their time,
- maintaining structure,
- maintaining personal status and self-worth,
- not feeling 'old', and,
- in some cases, people have identified wishing to avoid spending too much time at home with their spouse as the reason for staying in work.

<sup>&</sup>lt;sup>40</sup> Phillipson, Smith (2005)

<sup>&</sup>lt;sup>41</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>42</sup> Phillipson, Smith (2005)

<sup>&</sup>lt;sup>43</sup> Phillipson, Smith (2005)<sup>44</sup> Hedges, Sykes (2009)

#### Expectation of working at older ages in future is increasing

The expectation of being in work in future is highly correlated with actually being in work,<sup>45</sup> and the expectation of working after SPA has been increasing over the last decade (Table 3).

Table 3: Average <sup>46</sup> expectations of being in work after different age	es by
age and gender in 2002/3 and 2008/9,47 England	

Expectation of being	Gender/Age	2002/3	2008/9
in work after age 55	Women aged 50-54	66%	72%
Expectation of being	Men aged 50-54	55%	60%
in work after age 60	Women aged 55-59	36%	48%
	Men aged 55-59	56%	62%
Expectation of being	Men aged 60 to 64	26%	32%
in work <i>after age</i> 65			

Why are some people not working longer?

#### Some people retire voluntarily before State Pension Age, others retire early because they are 'pushed' by redundancy, ill-health or the need to provide care

People who leave work before SPA do so for a variety of reasons. Some of those who leave before SPA may feel that they have made sufficient financial provision to support themselves through retirement. However many people say they are compelled to leave work before SPA due to circumstances beyond their control such as health problems or the need to provide care for a family member. Some people may leave work before SPA because they are made redundant and are unable to find another job.

Although many reasons are given for not working longer there is little quantitative evidence with which to gauge which reasons are the most important, or the possible interactions between the reasons. However, many of the people who leave work before SPA for reasons beyond their control may not have made sufficient financial provision for their retirement and may suffer financially as a result.

Men are more likely to retire *voluntarily* before their SPA than women (as opposed to being compelled to by circumstances). In 2008/9, around 29% of men reported that they were 'retiring' in the five years before their SPA, compared to around 8% of women.<sup>48</sup>

Those who are most likely to retire voluntarily before SPA are:49

• Those in the highest wealth quintile, who are 2.2 times more likely than those in the middle quintile to retire voluntarily before SPA.

<sup>45</sup> Banks, Tetlow (2008)

<sup>&</sup>lt;sup>46</sup> Mean average

<sup>&</sup>lt;sup>47</sup> Crawford, Tetlow (2010) p. 73 Table 2A.13.

<sup>48</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>49</sup> Crawford, Tetlow (2010)

- Those with a DB pension, who are almost twice as likely to retire voluntarily before SPA than those with no private pension.
- Those with a long standing illness, who are over 50% more likely to retire voluntarily before SPA than those without, (though a large number of older people with a long standing illness reported leaving work for illness reasons and not as 'retiring'.)
- Those in Northern England, who are more likely to report being retired voluntarily than those in the South East.

The next sections consider the reasons for involuntary retirement. In general, involuntary retirement is more likely than voluntary retirement to be associated with negative outcomes in retirement, such as lower levels of well-being.<sup>50</sup>

### Health problems are one of the main, non-voluntary reasons for people to leave work before State Pension Age

Around 20% of men and women in England aged between 50 and 54 have a disability which limits their ability to work. By the time men and women are between ages 60 to 64, around 30% have a disability that limits their ability to work (Chart 6).

#### Chart 651



Some people with a work limiting disability continue to work at older ages. However, in order to continue working with a disability many people will need support from their employer, for example, through

50 Smith (2004)

<sup>51</sup> Crawford, Tetlow (2010) p. 68, Table 2A.7. rounded to nearest percent

reduced hours or a shift of responsibilities. Some workplaces are unable or unwilling to offer support to disabled employees. Therefore some people with a disability, who may be capable of working if supported, may not have access to suitable support and may therefore be compelled to leave work altogether.

Age, gender, ethnicity, occupation and location affect both the likelihood of having a work limiting disability and the likelihood of continuing to work while disabled:

- Work limiting disability is more common amongst older people in lower wealth quintiles, though the direction of causation is not clear (the deprivation associated with low wealth could lead to disability, conversely being ill during working life could lead to low wealth as earning and saving capability may have been constrained).<sup>52</sup>
- Work limiting disability is also more common amongst older people with low levels of education.<sup>53</sup>
- The likelihood of working while disabled varies:
  - Ø Older people with a work limiting disability in the North East are less than half as likely to be working than those in the South East.<sup>54</sup>
  - Ø Older women who have a work limiting disability are less likely to be in paid work than men.<sup>55</sup>
  - Ø Manual workers are more likely than those in other occupations to leave work due to health problems, either because the employer cannot offer them suitable alternative employment or because the employee does not feel capable of changing/shifting roles.<sup>56</sup>
- Occupation type affects the type of disability people experience: white collar workers are more likely to report that health problems at work are stress-related rather than physical.<sup>57</sup>
- In the UK, people from particular ethnic minority groups are more likely to suffer from health problems than white people. For example, Pakistani men are twice as likely as those in the general population to suffer from heart disease.<sup>58</sup> People from some ethnic minority groups may therefore be more likely to need to leave work at younger ages as a result of their health problems.

#### Some people who leave work due to a disability, may be able to return to work at a later date

Of people aged between 50 and 69 who reported a work limiting disability in 2004/5, 41% reported no longer having a work limiting

<sup>&</sup>lt;sup>52</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>53</sup> Crawford, Tetlow (2010)

<sup>54</sup> Crawford, Tetlow (2010)

 $<sup>^{55}</sup>$  Crawford, Tetlow (2010) however this may be also related to the fact that women over age 50 are generally less likely to be in work than men

<sup>&</sup>lt;sup>56</sup> Hedges, Sykes (DWP) (2009)

<sup>57</sup> Phillipson, Smith (2005)

<sup>&</sup>lt;sup>58</sup> Mai Sims, Vanderaa (2008) The differences in health outcomes for people from particular ethnic

minority groups are due at least in part to a lack of access to culturally appropriate care and support from healthcare providers

disability either in 2006/7, 2008/9 or both.<sup>59</sup> However, whether someone can return to work will depend on whether there are any jobs on offer, whether a potential workplace can offer support in terms of flexibility and training (if necessary), and the employee's willingness to return to work. In practice, it may be very difficult for an older person to return to work after they have left work due to disability, even if they recover from their disability. On the whole, people who leave work due to health problems (after age 50) are unlikely to re-join the labour market.<sup>60</sup> This is due to several factors, including the following: many older people leave work due to health problems or caring responsibilities which may hinder their return; the older someone is when they leave the labour market the less likely they are to feel motivated to return; and there are perceptions among those out of work that discrimination among employers will prevent their return.<sup>61</sup>

### Returning to work at older ages can be difficult and associated with receiving lower wages

For each additional year that people over age 50 are out of work, women are around 9% and men are around 24% less likely to return.<sup>62</sup> And if people do return to work after time out, this is often associated with a reduction in wage of around a quarter for those over age 50, and around 20% for those aged 25-49.<sup>63</sup> The prospect of lower wages may make returning to work after time out less attractive. For example, the smaller the gap is between wages and benefits, the more reluctant many jobcentre attendees are to return to work.<sup>64</sup>

### Some people leave work at or before State Pension Age because they do not feel there is enough financial incentive to remain

Leaving work before or at SPA is not necessarily a poor choice for everyone. For people who have made extensive financial provision or are entitled to a good pension, leaving work before SPA is both more common (for example, people with a DB pension are twice as likely to retire before SPA than people with no private pension<sup>65</sup>) and less likely to lead to economic deprivation.

However, some people who may benefit from remaining in work at older ages may still choose to leave work. Some people might not understand the potential financial benefits of staying in work or even believe that working after the SPA is prohibited.<sup>66</sup> But even those who are aware that you can work after SPA may feel that there are incentives to leave work, for example, some people believe that post-SPA income is too heavily

<sup>59</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>60</sup> Phillipson, Smith (2005), Banks, Tetlow (2008)

<sup>61</sup> Phillipson, Smith (2005),

<sup>62</sup> Banks, Tetlow (2008)

<sup>&</sup>lt;sup>63</sup> Phillipson, Smith (2005)

<sup>&</sup>lt;sup>64</sup> Phillipson, Smith (2005)

<sup>65</sup> Crawford, Tetlow (2010)

<sup>66</sup> DWP (2010b)

taxed to make work after the SPA rewarding.67 Many people also have limited knowledge regarding the potential benefits of state and private pension deferral,68 or a belief that they will not live long enough to make deferral worthwhile.<sup>69</sup>

On the whole, qualitative studies reveal very little understanding amongst the majority of people regarding pension planning and the potential benefits of working longer. However, there is some evidence that communicating with people about the potential benefits can make working longer seem more attractive.70

#### Many people leave work to provide care for a family member

As people age they are more likely to provide care for a family member (most often a partner or a parent). Women, and people from some ethnic minority groups,<sup>71</sup> are more likely to provide care at older ages with 14% of women in England aged 50-59, and 15% of women in England aged 60-64, reporting that they 'actively provided care in the last week' in 2008/9 (Chart 7).





67 Hedges, Sykes (DWP) (2009)

68 In 2008-09 around 2% of individuals aged between SPA and age 75 were in receipt of a state pension but had deferred it until some point above their SPA, Crawford, Tetlow (2010)

69 Hedges, Sykes (DWP) (2009)

70 Hedges, Sykes (DWP) (2009)

<sup>71</sup> Such as people from Bangladeshi and Pakistani backgrounds, Buckner, Yeandle (2006)

72 PPI analysis of English Longitudinal Study of Ageing (ELSA) WAVE 4 data

Providing care impacts people's ability to work; in 2008/9, 10% of women and 1% of men in England aged 50 and older who were not in employment were providing care.<sup>73</sup> The need to provide care can lead to an exit from the labour market because the need to provide care often means that people are no longer able to work regular hours or work too far from their home. Some carers report having found satisfaction from their job but felt that they were unable to continue working due to the demands of caring.<sup>74</sup> Some who wished to return to work while caring found that suitable work (e.g., with flexible hours or close enough to home)<sup>75</sup> was not available to them.<sup>76</sup>

#### Obstacles to older working can come directly from employers

In order for older workers to be able to engage in employment there needs to be an appetite from employers to recruit and retain older workers. Employers can facilitate older working by engaging with 'active ageing' policies and practices by, for example, actively recruiting and offering training to older workers, engaging positively with diversity policies, offering on-going career management and options for phased and flexible working and retirement.

Before the introduction of the Employment Equality Age Regulations (EEAR), relatively few organisations seemed to be engaged in active ageing policies.<sup>77</sup> However, there is evidence that the introduction of the EEAR and the more recent removal of the Default Retirement Age has resulted in some progress with regards to the recruitment and retention of older workers, and engagement with diversity and performance management policies.<sup>78</sup> While this suggests that at least in terms of formal employer policies and HR best practice organisations are removing barriers to older working, more evidence is needed of this being converted into practical changes in behaviour and attitudes.

### Trends and changes within private pensions could affect work and retirement patterns

The levels and types of private pensions held by individuals can affect retirement behaviour. Individuals with Defined Benefit (DB) pensions are more likely to retire at earlier ages than individuals with Defined Contribution (DC) pensions, or those without any private pension.<sup>79</sup> A typical DB scheme allows individuals to keep building up a pension until a certain pension age (generally age 65) but does not increase after that point, so there is less incentive to stay in work.

<sup>73</sup> Crawford, Tetlow (2010) p. 65, Table 2A.5. rounded to nearest percent

<sup>&</sup>lt;sup>74</sup> Vickerstaff *et. al.* (2009)

<sup>75</sup> Vickerstaff et. al. (2009)

<sup>76</sup> Vickerstaff et. al. (2009)

<sup>77</sup> Metcalf, Meadows (2006)

<sup>78</sup> Metcalf, Meadows (2010), McNair et al (2007)

<sup>&</sup>lt;sup>79</sup> Crawford, Tetlow (2010)

In contrast, for those saving into a Defined Contribution pension there are greater incentives to work longer. For example, working and saving longer could: increase the number of contributions that have been paid into the fund, increase the period over which investment returns are built up, and, if an annuity is purchased to convert the pension fund into an income stream, reduce the cost of an annuity due to the shorter expected period in payment.<sup>80</sup>

The rest of this chapter examines trends within private pensions and investigates how these could affect the options people have around work and retirement.

- Defined Benefit pensions are becoming less common in the private sector: Within the private sector, many Defined Benefit (DB) pension schemes are closing to new members, future accruals and, in some cases, selling their liabilities to third parties and closing their schemes altogether. Some DB schemes which are remaining open in the private sector are trying to reduce costs by lowering their accrual rate, basing final pension income on average salary rather than final salary, or, as a result of a change in Government policy, changing the index by which pension payments and deferred entitlement is uprated from the Retail Prices Index (RPI) to the Consumer Prices Index (CPI), which generally increases more slowly out of the two.
- **Public sector Defined Benefit pensions are undergoing change:** Within the public sector, DB schemes are undergoing changes which could result in higher contributions from scheme members and a move to basing final pension income on average salary rather than final salary, as well as potential future increases in the age at which people are expected to take their pension. Public sector pension payments (as well as valuations of deferred entitlement) have all been adjusted to uprate annually in line with growth in the CPI rather than the RPI.
- Rises in the Normal Pension Ages of people in DB pension schemes could lead to scheme members working longer: The minimum age at which people can take their private pension without a reduction in value, the Normal Pension Age (NPA), rose from 60 to 65 in many public sector occupational pension schemes during the last few years. There are Government proposals to link some public sector schemes NPAs to the State Pension Age (SPA) in the future. Some private sector schemes may also decide to link their NPA's to the SPA in the future.

<sup>&</sup>lt;sup>80</sup> Assuming that other factors that could affect the annuity rates (such as changes in life expectancies and interest rates) are not changed in the intervening period.

People with occupational pensions have historically left work at younger ages on average than people without them,<sup>s1</sup> in 2008/9 those with DB pensions were almost twice as likely to retire before SPA than those without pension provision.<sup>s2</sup> However, rises in the NPA could lead to many people in occupational pension schemes waiting longer to take their pension and working longer in order to provide themselves with income in lieu of a pension.

- More people will be saving in DC pension scheme in the future: Because many private sector DB schemes are closing, many people who may in the past have saved in a DB scheme will only have the option of saving in a DC scheme if they want a private pension. Autoenrolment will also lead to an increase in people saving in DC pensions when it begins in 2012. Active membership in DC schemes could grow to around 15 million by 2020 and around 17 million by 2050, compared to an estimated 5 million in 2008.<sup>83</sup>
- An increased number of people saving in DC pensions rather than DB pensions can lead to greater risks and lower pension incomes for some individuals: Saving in a DC pension scheme is inherently more risky for the individual than saving in a DB pension scheme. This is because in a DC scheme, the risk of market fluctuations causing fund depletion or poor investment return is carried by the individual, whereas in a DB scheme the risk is carried by the employer. Members of DC schemes also receive lower pension incomes on average than those in DB schemes partly due to the lower average level of contributions in DC schemes:
  - **Ø** In private sector occupational DC schemes, average employee contributions in 2010 were 2.7% of salary and average employer contributions were 6.2 % of salary.<sup>84</sup>
  - Ø In private sector occupational DB schemes, average employee contributions in 2010 were 5.1% of salary and average employer contributions were 15.8 % of salary.<sup>85</sup>

As a result of the reduction in coverage of DB schemes and the growth of DC schemes, future pensioners will face more uncertainty about their income in retirement. This could mean that some people may have to work for longer than expected in order to ensure sufficient income during working life and save for longer into a pension to ensure they have adequate incomes in retirement.

<sup>&</sup>lt;sup>81</sup> Phillipson, Smith (2005), this could be partly because NPAs have historically been lower than SPA and some people may have felt that their occupational pension provided sufficient income to live off of in retirement. Some people may also have been encouraged to leave at NPA by their organisations.
<sup>82</sup> Crawford, Tetlow (2010)

<sup>&</sup>lt;sup>83</sup> PPI (2009b), PPI Aggregate Model

<sup>&</sup>lt;sup>84</sup> ONS (2011a)

<sup>85</sup> ONS (2011a)

#### Conclusions

Employment rates have been increasing for people aged 50 and above:

- In 1993, around 64% of men aged 50 to 64 were in work. In 2011, this had increased to around 70% of men aged 50 to 64 in work.
- In 1993, around 57% of women aged 50 to 59 were in work. In 2011, around 72% of women aged 50 to 59 were in work.
- In 1993, around 8% of men over the SPA (65+) were in work. In 2011, this had increased to around 11% of men over the SPA (65+) were in work.
- In 1993, around 8% of women over the SPA (60+) were in work. In 2011, around 13% of women over the SPA (60+) were in work.

The use of flexible working, such as part-time working, as a way to phase into retirement is growing amongst older people though not all organisations are able or willing to offer flexible working to their employees.

Many people are compelled to leave work before SPA due to circumstances beyond their control such as health problems or the need to provide care for a family member. Some people with a disability or caring responsibilities, who may be capable of working if supported, may not have access to suitable support from their employer and may therefore be compelled to leave work altogether.

People who leave work before SPA for reasons beyond their control may not have made sufficient financial provision for their retirement and may therefore suffer financially as a result, and have lower levels of wellbeing.

In order for older workers to be able to engage in employment there needs to be an appetite from employers to recruit and retain older workers and to provide appropriate support to those older people who need it. While there is some evidence of this in HR policies, it is not clear if this has yet fed through into employer behaviour and attitudes.

As a result of the reduction in the coverage of Defined Benefit (DB) schemes and the growth of Defined Contribution (DC) schemes, future pensioners will face more uncertainty about their income in retirement. This could mean that some people may have to work for longer than expected in order to ensure sufficient income during working life and save enough in a pension to ensure they have adequate incomes in retirement.

# Chapter two: what are the implications of the Government's extending working lives policy agenda?

This chapter describes and examines the potential implications of Government policies which will affect future work and retirement patterns for older people.

### The Government is committed to encouraging people to work for longer

The Government is keen to encourage people to work longer in order to combat the potential financial problems both for the individual and for the state posed by a declining birth rate and increases in longevity. The Government has said that it *"would like to foster a climate in which people become: less inclined to retire before State Pension Age (SPA); more likely to give favourable consideration to working on after they reach SPA; [and] more understanding of any regulatory changes the Government may need to bring in to cushion the effects of pension shortfalls, where these may lead to people working longer."<sup>86</sup>* 

### What constitutes the Government's extending working lives policy agenda?

The Government has a range of policies and programmes aimed at extending working lives. The rest of this chapter describes these in turn and investigates what the implications might be for the Government and for individuals of the following policies:

- Policies aimed at moving people off benefits and into paid work:
   Ø Changes to disability related benefits
  - Ø The Government's Work programme
- Changes to age at which people can receive state pension and benefits income:
  - Ø State Pension Age rises
  - Ø Changes to the age at which people can claim Pension Credit
- Removal of barriers to working longer:
  - Ø Removal of the Default Retirement Age
  - **Ø** Efforts to tackle age discrimination in the workplace: The Employment Equality Age Regulations 2006; The Age Positive Initiative

## Policies aimed at moving people off benefits and into paid work

In order to tackle unemployment and ensure the sustainability of the benefit system, the Government has made several changes to the benefit system

The Government is committed to increasing employment amongst people of all ages and reducing benefit dependency in order to ensure the sustainability of the benefit system and to make sure that people who are able to work do so. Therefore they have made some changes to the way that disability benefits are paid and the way that 'fitness for work' is determined for benefit applicants. Specifically, Incapacity Benefit and disability related Income Support are being phased out and are being replaced by Employment and Support Allowance (ESA).

The Government has designed the ESA eligibility assessment to be stricter than the previous Incapacity Benefit assessment, so that fewer people will be eligible for means-tested benefits. The Government anticipates a reduction in benefit claimants of around 110,000 people by 2017/18 with yearly savings reaching £215m by 2017/18.<sup>87</sup>

### Under the new benefit assessment regime, fewer individuals are being found eligible for benefits

Under the previous Personal Capability Assessment used for Incapacity Benefit before the introduction of ESA, around 37% of claimants were found Fit for Work and therefore ineligible for benefits. A higher proportion of claimants have been found Fit for Work under the newer ESA work capability assessment where around 57% of new claimants were found Fit for Work (before taking appeals into account) and ineligible for ESA.<sup>88</sup>

As well as assessing all new claims for benefit under the new assessment regime, the Government has also been re-assessing existing recipients of Incapacity Benefit under the new regime, and adjusting benefit entitlement subject to the outcome of the assessment. There are differences in the outcomes of assessments between new and existing claimants. For claimants *transitioning from Incapacity Benefit to ESA*, 37% were assessed as Fit for Work and are not entitled to ESA (before taking appeals into account).<sup>89</sup>

<sup>87</sup> DWP (2008), p. 5

 <sup>&</sup>lt;sup>88</sup> DWP (2012b) official statistics for all new claims started between March 2011 and May 2011
 <sup>89</sup> DWP (2012a)
**Many older people will be affected by the changes to the benefit system** Around a million people who are over age 50 are currently receiving Incapacity Benefits.<sup>90</sup> At some point during the next few years these people will be phased on to ESA and it is likely that many will be found Fit for Work under the new, stricter assessment regime and therefore ineligible to continue receiving benefits unless they successfully appeal the decision. Some older people who are found Fit for Work may have difficulties finding work.

## Those who are found Fit for Work will not necessarily all end up in paid employment

For those aged 50 and over who have applied for the new Employment and Support Allowance and either were found Fit for Work, and therefore ineligible for benefits, or had dropped their claim, a year to 16 months later:<sup>91</sup>

- 28% of those aged 50–54 and 24% of those aged 55+ were in work,
- 23% of those aged 50–54 and 17% of those aged 55+ were temporarily unable to work due to sickness,
- 29% of those aged 50–54 and 30% of those aged 55+ were permanently unable to work due to sickness,
- 12% of those aged 50–54 and 11% of those aged 55+ were looking for work,
- 3% of those aged 50–54 and 1% of those aged 55+ were providing care.

For those who do not successfully appeal a Fit for Work decision, there are several potential destinations such as: entering paid employment, claiming another benefit, or living off a partner's income or off their savings.<sup>92</sup>

Those who are found Fit for Work but are unable to find employment may suffer from a reduction in income as well as from the stress of being unable to find work. It is well recognised that:

- It is harder for those who have been unemployed than those who have been in continuous employment to find work;<sup>93</sup>
- This difficulty increases the longer the period of unemployment lasts; and,
- Older people especially can find it difficult to return to work after a period of unemployment.<sup>94</sup>

<sup>92</sup> Barnes *et. al.* (2011)

<sup>&</sup>lt;sup>90</sup> DWP tabulation tool, http://83.244.183.180/100pc/tabtool.html, (February 2011 data) includes Severe Disablement Allowance recipients (closed to new claimants). Existing recipients of Severe Disablement Allowance are being phased on to ESA alongside Incapacity Benefit recipients. <sup>91</sup> Barnes *et. al.* (2011) figure 4.1, excludes successful appeals and appeals in progress

<sup>&</sup>lt;sup>93</sup> Because of many factors including: the attitudes of employers to people who have spent time outside the workforce; mental health problems and other behavioural and emotional effects of unemployment; deterioration of skills; deterioration of confidence etc.

<sup>94</sup> Hedges, Sykes (DWP) (2009)

Those who were found Fit for Work but did not find work were more likely to experience deterioration in health and well-being,<sup>95</sup> making it less likely that they would be able to work in the future.

### The Work Programme is intended to provide help to the long-term unemployed and vulnerable groups

Alongside changes to benefits, the Government has also introduced the Work Programme,<sup>96</sup> which is intended to provide help to the long-term unemployed who have not been able to find work through Job Centre programmes and assistance.

Many older people who may be moving off benefits or transitioning from Incapacity Benefit to ESA may end up in the Work Programme as they may find it harder to return to work than younger participants.

The Work Programme was launched in 2011 and there is not yet enough data to assess the impact it is having on older people and their employment. An evaluation of the Work Programme is due to be undertaken between 2011 and 2013<sup>97</sup> which will give an indication of the impact that the Work Programme is having on the employment of older people.

# Changes to the age at which people can receive state pension and benefits income

### Increases in longevity can increase Government expenditure on the state pension

When the State Pension Age (SPA) of age 65 for men and age 60 for women was introduced in 1948 the life expectancy of a man aged 65 was 12 years and the life expectancy of a woman aged 65 was 15 years.<sup>98</sup> However, in 2011 the life expectancy of:

- a man aged 65 had risen to 21 years, and is expected to rise to 22 years by 2015 and 25 years by 2050.99
- a woman aged 65 had risen to 24 years, and is expected to rise to 25 years by 2015 and 28 years by 2050.<sup>100</sup>

### Increases in longevity increase the cost to the taxpayer of paying the State Pension

The Government estimates that the UK is currently spending around £100 billion per year on 'pensioner benefits' which include the state pension, Pension Credit and other benefits for pensioners.<sup>101</sup> This is around 6.9% of

<sup>95</sup> CAB (2010)

 $<sup>^{96}\,</sup>www.dwp.gov.uk/policy/welfare-reform/get-britain-working/\#work$ 

<sup>97</sup> By a consortium led by the Institute for Employment Studies, www.employment-

studies.co.uk/main/index.php

<sup>98</sup> Office of Health Economics data

<sup>&</sup>lt;sup>99</sup> ONS 2010 based cohort expectations of life, www.ons.gov.uk

 $<sup>^{\</sup>rm 100}$  ONS 2010 based cohort expectations of life

<sup>&</sup>lt;sup>101</sup> DWP (2010a)

GDP in 2010.<sup>102</sup> The proportion of GDP spent on the state pension and pensioner benefits will rise as longevity increases, unless either the SPA rises with it or the Government reduces the amount people receive from the state pension. The Government estimates that, as a result of upwards revisions of life expectancy, they may have to spend around £6.5 billion more than previously estimated on the cohort of pensioners who reached SPA in 2010 over their lifetime.<sup>103</sup>

### The Government is raising the age at which people can claim their state pension

Without policy change, increases in longevity result in people receiving their state pension for longer periods than older generations. For example, people who reached SPA in 2010 will on average receive their state pension for 32% of their adult life, compared to those who reached SPA in 1981, who will receive their state pension for, on average, 25% of their adult life.<sup>104</sup>

If the SPA remains unchanged and longevity continues to improve, the proportion of adult life spent in receipt of the state pension would continue to increase. This would not only increase the cost to the UK taxpayer and the Government of providing state pensions, it also creates intergenerational unfairness because it means that the level of benefit people receive from the state pension depends on the generational cohort to which they belong.

Therefore, the Government is raising the SPA. The SPA in the UK in 2012 is age 61 for women and is age 65 for men. From 2010 to 2018, women's SPA is rising and will reach age 65 by 2018. Both men and women's SPA will rise to 66 by 2020. Current legislation means that the SPA will continue to rise further in a series of steps: from 66 to 67 between 2026 and  $2028^{105}$  and from 67 to 68 between 2044 and 2046 for both men and women (Chart 8).<sup>106</sup> The Government estimates that their revisions to the SPA rise timeline will save £30.6 billion in spending on state pensions between 2016/17 and 2025/26.<sup>107</sup>

<sup>104</sup> Adult life is assumed to begin at age 18 and assuming a combined average life expectancy for men and women. PPI Briefing Note 60 *What are the implications of the Government's latest legislation increasing State Pension Age*? www.pensionspolicyinstitute.org.uk

<sup>105</sup> Pensions Act 2011

<sup>&</sup>lt;sup>102</sup> 2010 GDP = £1,458,452,000,000 ONS Blue Book 2011, www.ons.gov.uk <sup>103</sup> DWP (2010a)

<sup>&</sup>lt;sup>106</sup> Pensions Act 2007

<sup>&</sup>lt;sup>107</sup> DWP (2011b) Table 3



The current Coalition Government has indicated that it believes that the SPA may need to rise faster and further than this, that there may need to be a link to longevity in future increases (to be reviewed automatically when life expectancy projections are revised), and that it might bring forward the rise to age 68.<sup>108</sup>

The next section of this chapter briefly summarises the impact that SPA rises could have on the incomes of particular groups, specifically those who may have either shorter life expectancies or those who might need to leave work at earlier ages.

### State Pension Age rises might reduce the proportion of life people in lower social classes spend in receipt of the state pension relative to those in higher social classes

People in lower social classes have lower life expectancies on average than those in higher social classes.<sup>109</sup> Because of this, any rises in SPA will mean that those in lower social classes and manual jobs will spend, on average, relatively less time in receipt of the state pension as a result of rises in SPA than those with higher life expectancies.

The Government argues that the current legislated rises in SPA will not unfairly disadvantage those with lower life expectancies because people in all social groups will still experience increases in life expectancy which

<sup>108</sup> DWP (2010a)

<sup>&</sup>lt;sup>109</sup> Using the Registrar General's classification of social class: I – Professional, II – Intermediate, IIN – Skilled non-manual, IIIM – Skilled manual, IV – Semi-skilled manual, V – Unskilled manual

are equal to or higher than the increases in SPA.<sup>110</sup> However, the *proportion* of life spent in receipt of state pension for those with lower life expectancies may be more substantially reduced by SPA rises than for those with higher life expectancies.

Those on lower incomes are likely to have lower private pension entitlement or savings when they reach retirement and are more likely to receive a large proportion of their income from the state pension. Therefore, they might lose a higher proportion of total lifetime income as a result of SPA rises than those with higher private pension savings and other income and assets.<sup>111</sup>

### Women will see a greater reduction in state pension income and in proportion of time spent in receipt of state pension as a result of equalisation

The SPA rises which are currently legislated for, especially the equalisation of men and women's SPA at age 65 by 2018, mean that women will lose out more than men both in terms of reduction of their total pension income (including Pension Credit for those on very low incomes) and in terms of a reduction in the proportion of life spent in receipt of state pension.

Women will be particularly affected by the equalisation of their SPA with men's between 2010 and 2018, which will be followed quickly by a rise to age 66 for both men and women by 2020. This involves women's SPA rising by 6 years over a period of around 10 years. The final rise has been accelerated by recent Coalition Government legislation. Women's SPA was originally intended to reach 65 in 2020 and 66 in 2028 alongside men's. The recent legislation has a particular impact on women who are currently within 10 or less years of their previously expected SPA who may have already made work, saving and retirement decisions based on having a particular SPA and who may not be able to adjust to a higher SPA by working or saving longer (see Box 2).

However, if women work longer as a result of SPA increases they may have a higher overall income in both working life and retirement. For some women who are able to work longer, SPA rises could therefore potentially be financially beneficial.

#### Box 2: Adjusting to rises in State Pension Age

One important consideration when evaluating when SPA should rise is the amount of notice people might need in order to adjust their finances An adequate amount of notice regarding SPA rises is necessary in order to allow people to adjust their working and saving plans accordingly. For people who are already economically inactive when an SPA rise is announced it may be difficult to adjust their work and retirement plans. This is because people may have saved up only enough to support themselves up until the SPA that they had on leaving paid work. It might be difficult for older people to re-enter the labour market in order to increase their retirement savings or fund themselves until their SPA.

SPA changes in the past have aimed to avoid this issue by using long lead-in times. Women were given at least 15 years' notice in relation to the equalisation of male and female SPAs between 2010 and 2020 as originally scheduled.<sup>112</sup>

Evidence on participation rates at older ages would suggest that if less than five years notice is given of changes to male SPA, many men might find it difficult to adjust. Within 5 years of the current SPA of 65, only 60% of the male workforce is still economically active. Men may need at least 5 years notice and ideally 10 years notice of an SPA change, because within 10 years of the current SPA of 65, around 75% of men were still economically active in 2010<sup>113</sup> and may therefore have been in a position to respond to the policy change by delaying their retirement if they needed to. A smaller proportion, around 65% of women aged 50-59 were economically active in 2010<sup>114</sup> suggesting that women may need more notice of SPA changes than men.

#### Women are likely to receive higher state pensions in future

Women have historically tended to have lower state pension entitlement than men, due to being more likely to take time out of work for caring responsibilities and because women have lower earnings on average than men. But in the future, (by around the end of 2018) around 90% of both women and men are expected to reach retirement with a full Basic State Pension.<sup>115</sup> Therefore, many women will be reaching SPA later than previous cohorts, but some will receive a higher state pension income as a result of reforms to the qualifying system, and the introduction in 2016 of the single-tier pension.<sup>116</sup>

 $<sup>^{112}</sup>$  The legislation for previous SPA rises was introduced in 1995 but the actual changes did not start to be implemented until 2010 to 2020

<sup>&</sup>lt;sup>113</sup> ONS, Pension Trends, Chapter 4 (2011)

<sup>&</sup>lt;sup>114</sup> ONS, Pension Trends, Chapter 4 (2011)

<sup>115</sup> DWP (2010a)

<sup>&</sup>lt;sup>116</sup> Announced in the Chancellor of the Exchequer's Budget Statement, March 12, 2012 www.hm-treasury.gov.uk

People from some ethnic minority groups and people with disabilities are less likely to be in paid work and may therefore be more affected by State Pension Age rises<sup>117</sup>

People from non-white ethnic groups are more likely to have low or no private pension savings, more likely to be in receipt of state benefits during working life and twice as likely as white people to be entitled to Pension Credit when they reach age of entitlement.<sup>118</sup> People from some ethnic minority groups are also more likely to be unemployed during working life, especially people from a black ethnic background. People from Asian ethnic backgrounds are less likely to be in paid work than other ethnic groups, due to illness, disability or the need to provide care.<sup>119</sup>

Therefore rises to SPA are likely to have a greater negative financial impact on people from some ethnic minority groups than they will on the general population. Similarly, people with disabilities are less likely to be in paid work and more likely to be eligible for Pension Credit at the minimum qualifying age, and may therefore lose out more than people in other groups as a result of policy changes.<sup>120</sup>

### Changes to the age at which people are entitled to Pension Credit could affect the incomes of some people over the age of 60

Pension Credit is the main means-tested benefit for pensioners and there are currently around 2.5 million pensioners claiming it.<sup>121</sup> Pension Credit is made up of two elements: Guarantee Credit, which tops pensioners' income up to a minimum weekly income of £142.70 (2012/13) and Savings Credit, which pays extra income to pensioners with savings.<sup>122</sup>

The age at which pensioners can claim Pension Credit is linked to women's SPA and has therefore been age 60 since its introduction in 2003. Changes to women's SPA mean that the age at which all people can claim Pension Credit is rising, and could impact the incomes of people between age 60 and SPA who may previously have been able to claim Pension Credit. The age at which some people can claim Pension Credit will also be impacted by changes in the Welfare Reform Act 2012 which stipulate that an individual over the SPA will not be entitled to Pension Credit if they have a spouse under the SPA. This stipulation will come into force when Universal Credit is introduced, which is currently expected to be in October 2013.

<sup>117</sup> There isn't enough data on life expectancy and ethnicity to make conclusions regarding how SPA rises might affect the proportion of lifetime that people from ethnic minority groups spend in receipt of the state pension

<sup>118</sup> DWP (2010a)

<sup>119</sup> DWP (2010a)

<sup>120</sup> DWP (2010a)

<sup>121</sup> DWP data, February 2011. DWP tabulation tool, http://83.244.183.180/100pc/tabtool.html
 <sup>122</sup> Including state and private pensions in payment

### The Government is investigating options for linking future rises in State Pension Age to longevity

If longevity continues to increase, the SPA will also need to continue rising in order to ensure that the state pension remains affordable and that there is intergenerational fairness. The Government has announced an intention to ensure the SPA is increased in future to take into account increases in longevity.<sup>123</sup> Although the Government has not yet determined the formula they will use, they have suggested that there will be automatic reviews of SPA whenever there is a revision of life expectancy projections. The Government plans to publish proposals later in 2012.<sup>124</sup>

### Many countries are exploring ways of linking pensions to life expectancy, but linking Pensionable Age (State Pension Age) to life expectancy is still relatively uncommon

Many countries are exploring methods of linking state pension benefits to life expectancy; either by adjusting the amount of benefit that people are paid or by adjusting the number of qualifying years required to be eligible for a full state pension. However, linking Pensionable Age itself to life expectancy is less common; only Denmark, Italy and Greece are currently legislating for a direct link between Pensionable Age and life expectancy.<sup>125</sup>

The following case study describes Denmark's plans for linking life expectancy to their Pensionable Age (State Pension Age) and highlights the lessons that the Danish experience has for the UK.

### International Case Study: Denmark - Linking Pensionable Age to life expectancy

### Denmark - relevant statistics:126

- **Pensionable Age** 65, rising to 67 by 2022 and 68 by 2030 (UK equalising at 65 then rising to 66 by 2020, 67 by 2026, then 68 by 2046 at the latest)
- Average life expectancy at age 65 83127 (UK 84)128
- % GDP spent on state pension 5.6% (UK 6.9%)
- Average age of labour market exit for men (2011) 64 (UK 64)<sup>129</sup>

#### Denmark will link Pensionable Age to life expectancy from 2015

From 2015, Danish Pensionable Age will be indexed to life expectancy using the average life expectancy of 60 year olds over the previous two

<sup>129</sup> OECD (2011)

 <sup>&</sup>lt;sup>123</sup> Chancellor of the Exchequer's Budget Statement, March 12, 2012 www.hm-treasury.gov.uk
 <sup>124</sup> At the time of the Office for Budget Responsibility's (OBR) 2012 Fiscal sustainability report
 <sup>125</sup> OECD (2011)

<sup>&</sup>lt;sup>126</sup> OECD (2011) OECD figures are used in order to make comparisons across countries on a consistent basis, internal country statistics may vary from OECD stats.

<sup>127</sup> OECD figure of average between men and women.

 $<sup>^{\</sup>rm 128}$  OECD (2011) this is the OECD figure for UK life expectancy.

years. Under current calculations, Pensionable Age is expected to rise from 65 to 67 by 2022, and reach 68 by 2030, 70 by 2040, and 72 by 2055 (Chart 9).

In the UK under current legislation the SPA will rise from just above age 60 for women to age 65 between 2010 to 2018, and both men and women's SPA will rise from 65 to 66 by 2020. The SPA will continue to rise further in a series of steps: from 66 to 67 between 2026 and 2028 and from 67 to 68 between 2044 and 2046 for both men and women (Chart 9).

### Denmark is reducing the average time spent in receipt of the state pension in order to ensure sustainability

The average number of years Danish people spend in receipt of their state pension, in 2010, is around 19½ years. The Government is raising the Pensionable Age in a series of steps in order to reduce the average number of years Danish people spend in receipt of their state pension to 14½ years, and then maintain that average with increases linked to life expectancy. The Danish Government anticipates that the proportion of the population in work will increase as a result of policies on work, disability and changes to Pensionable Age.<sup>130</sup> Chart 9 shows the projected impact on the Danish Pensionable Age of both improvements in life expectancy and the gradual reduction in the average number of years people will receive their state pension.



<sup>130</sup> PM Helle Thorning-Schmidt speech to Parliament October 2011 www.stm.dk/\_p\_13523.html
 <sup>131</sup> Danish ministry of finance

#### There are safeguards built into the system

All Danish employees, as well as the majority of people on benefits and many self-employed people belong to ATP (the Danish Labour Market Supplementary Pension). ATP is a *"statutory, fully funded, collective insurance based, defined-contribution scheme."*<sup>132</sup> ATP directly pays a flat-rate pension annuity to its members from the age of 65 and functions as an addition to the Danish state pension.

ATP allows people who have made sufficient contributions and are unemployed or working part-time to make use of an *early retirement age* and take their ATP pension up to five years before Pensionable Age at a reduced rate. The early retirement age is being reduced in a series of stages to three years before Pensionable Age between 2018 and 2023. There is an exception for those who are disabled who can claim an early ATP pension on the basis of their disability.

There are therefore some safeguards built into the rises to Pensionable Age in the form of the early retirement and the exception for disabled people, though they will not protect people who have not made sufficient contributions (such as some self-employed people) who might need to claim state benefits if they cannot work anymore.

#### Conclusions and lessons for the UK from the Danish experience

Denmark is raising their Pensionable Age in line with projected changes in life expectancy from 2015 in order to ensure sustainability of their state pension. Though these rises will ensure that the Danish Government can continue to pay the state pension without reducing the benefits paid the rises also mean that some people who cannot work until Pensionable Age will have to wait longer for their state pension. There are some safeguards built into the system in the form of the early retirement age and the special senior disability scheme, although the minimum age for early retirement is rising alongside changes to Pensionable Age.

The current Coalition Government has announced that there will be a link between the SPA in the UK and life expectancy in future. The UK SPA may therefore rise fairly quickly, as seen in current projections of rises in the Danish Pensionable Age. If the UK were to adopt a direct link between SPA and life expectancy then safeguards may be needed to protect those who may not be able to work longer.

<sup>132</sup> OECD (2011)

### The removal of barriers to working longer

## The Government has removed the right to compulsorily retire employees on the basis of their age

In 2006, the previous Labour Government brought in the Default Retirement Age (DRA). The DRA allowed employers to compulsorily retire employees when they reached the age of 65 while also removing the power of employers to compulsorily retire employees at ages *below* 65 which had previously been legal, although exceptions applied for people in some occupations.

In 2011, the Coalition Government fully removed the right to compulsorily retire employees on the basis of their age, by abolishing the DRA.<sup>133</sup>

### Removing the DRA will allow some employees to continue to work after age 65 if they wish to, however many employees already had that option

While the removal of the DRA will allow some employees to continue working past the age of 65 many employees may have retired voluntarily at, or under, age 65 regardless of the presence of the DRA. Furthermore, not all employers made use of the DRA before it was removed; less than half the workforce worked in an organisation with a compulsory retirement age before the abolition of the DRA.<sup>134</sup>

Even when it was legal to compulsorily retire employees at age 65, people were allowed to request to work on beyond age 65 and around 80% of requests to stay on were accepted by employers.<sup>135</sup> In 2009, according to research commissioned by Age UK, around 100,000 people were made to retire at or over the age of 65 under DRA legislation.<sup>136</sup> Therefore, the removal of the DRA is likely to affect a minority of people who would like to work after age 65 but whose employers might not have allowed them to under the previous system. However, it is possible that in future many more employees will wish to work beyond the age of 65 than do so today. In future, the absence of the DRA may remove a barrier for a larger number of employees who wish to or need to work for longer.

The USA removed the right to compulsorily retire employees on the basis of age during the 1970s and 1980s. The following case study investigates the effects this had on employment rates and highlights the lessons that the US experience may have for the UK.

<sup>133</sup> The Employment Equality (Repeal of Retirement Age Provisions) Regulations 2011
<sup>134</sup> Metcalf, Meadows (2010)
<sup>135</sup> Sykes *et. al.* (2010)
<sup>136</sup> AGE UK (2010)

# International Case Study: USA - 25 years after the removal of mandatory retirement ages

USA - relevant statistics:137

- **Pensionable Age** 66 from 2008, rising to 67 by 2022 (UK equalising at 65 then rising to 66 by 2020, 67 by 2026, then 68 by 2046 at the latest)
- Average life expectancy at age 65 84 (UK 84)<sup>138</sup>
- % GDP spent on state pension 6% (UK 6.9%)
- Average age of labour market exit for men 66 (UK 64)<sup>139</sup>

### The USA removed mandatory retirement in a phased process between the late 1970s and 1986

In 1986 the USA became one of the first countries to outlaw mandatory retirement on the basis of age,<sup>140</sup> as a result of pressure from lobbying groups, a desire to tackle age discrimination in the workplace and the financial pressures on the state pension funding as a result of an ageing population.<sup>141</sup> In the 1960s and 1970s between 40% and 50% of men in the USA worked for an employer who had a mandatory retirement age.<sup>142</sup>

**Removing the option for employers to use mandatory retirement ages increased employment for those aged 65 to 70 by between 10% and 20%** Research which allowed for other factors affecting labour force participation in the USA, shows that labour market participation by those aged 65 to 70 rose by between 10% and 20% during the 1980s as a direct result of the phasing out of Mandatory Retirement.<sup>143</sup> There is evidence that abolishing Mandatory Retirement has somewhat smoothed out the historical spike of retirement at age 65.<sup>144</sup>

### Some employers made changes to their pension arrangements to encourage older workers to retire

There is some evidence that after the removal of Mandatory Retirement, employers in the USA initially started using private pensions to encourage employees to retire by lowering Normal Pension Ages and increasing the generosity of early retirement options.<sup>145</sup> However this changed during the late eighties as a result of economic pressures on companies that provide pensions to their employees, and employers began to reduce the generosity of early retirement options.<sup>146</sup>

<sup>&</sup>lt;sup>137</sup> OECD (2011) OECD figures are used in order to make comparisons across countries on a consistent basis, internal country statistics may vary from OECD stats.

<sup>&</sup>lt;sup>138</sup> OECD (2011)

<sup>139</sup> OECD (2011)

 $<sup>^{\</sup>rm 140}$  Exceptions applied to certain employees for example, those in the armed services, aviation, judicial and safety industries

<sup>&</sup>lt;sup>141</sup> Wachter (2009)

<sup>&</sup>lt;sup>142</sup> Wachter (2009)

<sup>&</sup>lt;sup>143</sup> Wachter (2009)

<sup>&</sup>lt;sup>144</sup> Wachter (2009)

<sup>&</sup>lt;sup>145</sup> Wachter (2009) <sup>146</sup> Wachter (2009)

### Pension arrangements continue to affect retirement decisions in the USA

Changes to pension arrangements are continuing to impact work and retirement decisions in the USA: in particular the shift from Defined Benefit to Defined Contribution pensions is changing the way many people access their private pension savings<sup>147</sup> and the amount of income people expect to receive. This means that some people may be working longer in order to ensure they have a sufficient pension or other savings and assets to finance their retirement.

### Pension arrangements are only one among a number of factors impacting the decisions people in the USA make about retirement

The employment rate among older people in the USA has been growing since the 1980s (Chart 10) due to many factors, including the removal of mandatory retirement, but also because of trends similar to those experienced in the UK, such as the shift from DB to DC pensions, increases in longevity and healthy longevity and the trend for more people to need to work longer from economic necessity.<sup>148</sup>



<sup>147</sup> Anderson et. al. (1997)

<sup>148</sup> Boston College, Center for Retirement Research (2008) A Micro-Level Analysis of Recent Increases in Labour Force Participation Among Older Workers http://crr.bc.edu/

<sup>149</sup> Data from United States Department of Labor, Bureau of Labor statistics using data from the Current Population Survey, www.bls.gov

#### Conclusions and lessons for the UK from the US experience

Lifting the mandatory retirement age has contributed to increases in employment at older ages in the USA. However there are many other factors also playing a role. Economic changes, changes to pension arrangements and increasing the Pensionable Age are likely to also contribute to current and future increases in employment at older ages. The USA has put in place safeguards which enable individuals to take their state pension early, at a reduced rate.

# Efforts to tackle age discrimination in the workplace in the UK

### 2006 legislation tackling age discrimination could make working at older ages easier for some employees

Age discrimination can manifest itself through the decisions employers make regarding recruitment, promotion, pay rises, redundancies, offers of training, supervision and other support, and the general treatment of older people in the workplace. Around 40% of older employees report having experienced age related discrimination at work.<sup>150</sup> Recognising that age discrimination represents a barrier to the success of a policy agenda which aims to extend working lives, the previous Labour Government brought in legislation designed to tackle age discrimination in the work-place in 2006, (the Employment Equality Age Regulations 2006 (EEAR)) alongside the introduction of the DRA.

### EEAR and the introduction of the DRA have led to some change among employers regarding redundancies and performance management policies

There is no available data on the effect of the EEAR on redundancies for workers over the age of 65 as the DRA was still in force during the most recent surveys measuring the impact of the EEAR. However, the EEAR and the introduction of the DRA do appear to have had some effect on reducing redundancies linked to ages below 65, and has prompted growth in formal performance management policies and equal opportunities policies.<sup>151</sup> Around 32% of organisations, employing 55% of the workforce, were found to have changed some of their policies or practices as a result of the EEAR. Positive age related changes were least common in smaller organisations, private sector organisations and the construction industry.<sup>152</sup>

### The Government has introduced an initiative aimed at encouraging employers to adopt more age positive policies

In 2001, as part of the wider policy strategy aimed at extending working lives and enabling people who are dependent on state benefits to return to employment (the Welfare to Work campaign) the previous Labour

<sup>&</sup>lt;sup>150</sup> Macleod *et. al.* (2010)

<sup>&</sup>lt;sup>151</sup> Metcalf, Meadows (2010)

<sup>&</sup>lt;sup>152</sup> Metcalf, Meadows (2010)

Government introduced the Age Positive Initiative. The Age Positive Initiative is the responsibility of the DWP and aims to "raise employer's awareness and change attitudes and culture through highlighting the business benefits of an age diverse workforce."<sup>153</sup>

The Age Positive initiative focuses mainly on providing sector bodies with guidance and research in order to encourage them to take Age Positive codes of practices, initiatives and policy documents forward under their own branding. For example, in 2009 the Institute of Directors published 'A Director's Guide: Extending working lives. How businesses can benefit from the skills of an ageing workforce',<sup>154</sup> which drew on Age Positive information and support.

## There is still age discrimination in employment, and some employers may have to work to ensure they are complying with the EEAR

Despite the EEAR and the Age Positive Initiative, discrimination at older ages continues to remain a barrier to work for older employees, especially in the areas of recruitment and offers of training and skill development. This could be the result of a lack of awareness among managers regarding their own organisations policies, and can manifest itself through a lack of awareness or understanding on the part of managers and employers of the needs of older people, or a lack of willingness on the part of employers to adapt, for example, by offering more flexible working options.<sup>155</sup>

### New Zealand implemented an extending working lives policy approach similar to the one currently being deployed in the UK

New Zealand implemented a policy approach to extending working lives which contained policies that are similar to the ones being deployed in the UK: raising the State Pension Age (SPA), removing default retirement and attempting to tackle barriers to working at older ages. The following case study investigates the effects these policies had on employment rates and the labour market, and highlights the lessons that the New Zealand experience has for the UK.

<sup>153</sup> http://statistics.dwp.gov.uk/asd/asd5/agepositive.asp
 <sup>154</sup> www.director.co.uk/content/pdfs/EWL\_Jul09.pdf
 <sup>155</sup> Macleod *et. al.* (2010)

# International Case Study 3: New Zealand - A three strand policy approach to extending working lives

New Zealand - relevant statistics:156

- **Pensionable Age** 65 (UK equalising at 65 then rising to 66 by 2020, 67 by 2026, then 68 by 2046 at the latest)
- Average life expectancy at age 65 84 (UK 84)<sup>157</sup>
- % GDP spent on state pension 4.3% (UK 6.9%)
- Average age of labour market exit for men 67 (UK 64)<sup>158</sup>

### New Zealand instituted a multi-strand policy approach to extending working lives in the 1990s, similar to the policy approach being pursued in the UK

New Zealand began to tackle the issue of the need to extend working lives before many other countries as part of a drive to ensure the sustainability of the state pension. The three-pronged policy approach that New Zealand used in the 1990s is similar to the one the UK is currently implementing. The policy approach in New Zealand included the following:

- Mandatory retirement on the basis of age was made illegal in a staged process between 1993 and 1999.
- The New Zealand Pensionable Age (the age at which people are eligible to claim a full state pension) was increased from age 60 to age 65 between 1992 and 2001.
- The removal from 1998 of taxation on non-state pension income (which had been in effect since 1985).
- The New Zealand Positive Ageing Strategy was introduced in 2001 and focussed specifically on eliminating ageism in the workplace and the promotion of flexible work options to enable older people to work for longer.<sup>159</sup>

#### Working at older ages has been increasing in New Zealand

Since the three policies were implemented in the 1990s, working at older ages has increased substantially in New Zealand. People who are aged 60 to 64 have seen the most dramatic increase in participation rates probably owing, at least partially, to the rise in Pensionable Age from age 60 to age 65. In 1990, 26% of people aged 60 to 64 in New Zealand were in employment. By 2000, this had risen to 46% and by 2010, 70% of people aged 60 to 64 in New Zealand were in employment (Chart 11).<sup>160</sup>

<sup>&</sup>lt;sup>156</sup> OECD (2011) OECD figures are used in order to make comparisons across countries on a consistent basis, internal country statistics may vary from OECD stats.

<sup>&</sup>lt;sup>157</sup> OECD (2011)

<sup>&</sup>lt;sup>158</sup> OECD (2011)

 $<sup>^{159}\</sup> www.msd.govt.nz/about-msd-and-our-work/publications-resources/planning-strategy/positive-ageing/goals-and-actions.html$ 

<sup>&</sup>lt;sup>160</sup> New Zealand Government statistics using the Household Labour Force Survey (HLF), Annual December data, www.stats.govt.nz/infoshare/Default.aspx

Labour market participation for people over age 65 has increased as well, at least partially as a result of rises in the SPA, the removal of mandatory retirement and changes in employment policy.<sup>161</sup> However other factors have played a significant role in increasing participation at older ages such as: a skills shortage facing employers; a decrease in the taxes charged for those who earn income as well their pension; improvements in health; an increase in the availability of work which is not too physically taxing; and the incentive to work inherent in a system with little means-testing.<sup>162</sup> In 1990, 7% of those over age 65 were in employment, which rose to 17% by 2010 (Chart 11).



#### The policy changes have prompted New Zealand employers to offer more flexible working options to older employees

One of the impacts of the extending working lives policy approach in New Zealand has been an increase in options for flexible working. Flexible transitions into retirement have become common in New Zealand, with around 75% of those in their early to mid-sixties experiencing some form of phased transition into retirement, many of whom went from full-time to part-time work before retiring.<sup>164</sup>

<sup>161</sup> New Zealand Department of Labour (2007)

<sup>163</sup> New Zealand Government statistics using the Household Labour Force Survey (HLF), Annual December data, www.stats.govt.nz/infoshare/Default.aspx
 <sup>164</sup> Dixon (2008)

<sup>&</sup>lt;sup>162</sup> New Zealand Department of Labour (2007)

### Conclusions and lessons for the UK

New Zealand began to tackle the issue of the need to extend working lives before many other countries. The three strand policy approach that New Zealand used in the 1990s is similar to the one the UK is currently implementing. Since the three policies were implemented in the 1990s working at older ages has increased dramatically in New Zealand. One of the impacts of the extending working lives policy agenda in New Zealand has been an increase in options for flexible working.

However, economic factors also contributed to the high rates of working at older ages in New Zealand, in particular, the skills shortage. The UK has a more open approach to immigration, meaning that employers in the UK may have more access to workers when skills shortages arise. Therefore it is not clear that the UK would see such dramatic improvements in the proportion of older people in work.

#### Conclusions

The Government has implemented a range of policies and programmes aimed at extending working lives, including

- policies aimed at moving people off benefits and into paid work;
- changes to the age at which people can receive the state pension and benefits income;
- the removal of barriers to working longer, including the removal of legal barriers and efforts to tackle age discrimination in the workplace.

While all of these policies will encourage and enable some people to work longer and take their state and private pension later, there may be negative financial implications for those who cannot work longer because of health problems, or other factors. For those who are unable to work longer, rises to the State Pension Age and changes to the benefit system may result in lower incomes in retirement.

### Chapter three: how much longer might individuals need to work and save in order to meet target levels of retirement income?

This chapter analyses how working and retirement patterns could impact on future levels of income adequacy for people in retirement.

### How long would people need to work in order to achieve adequacy in retirement?

This analysis investigates how long people aged between 50 and State Pension Age (SPA) in 2011 and still in work,<sup>165</sup> might need to continue working and saving at current levels in order to meet two different measures of income adequacy:

- The **Minimum Income Standard**, which calculates how much income pensioners require to meet a *minimum acceptable standard of living*,<sup>166</sup> is just under £11,000pa (£211pw) for a single pensioner Before Housing Costs (BHC), and around £15,700pa (£303pw) for a couple in 2011.<sup>167</sup> For people who have lived on medium to high incomes during working life, the minimum acceptable standard of living may not seem adequate as it would generally constitute a drop in living standards.
- Working life replacement rates which calculate how much income an individual pensioner might need in order to achieve a similar standard of living to the one they had in working life, are generally in the range of 50% to 80% of people's gross working life income. Lower income individuals have higher replacement rates than higher income individuals as their income may be closer to the levels required to meet basic needs. A median-earning man with a weekly income at the point of retirement of around £500pw<sup>168</sup> might need a gross weekly retirement income of around £17,400pa (£335pw) to meet a 67% replacement rate of working life income (and recreate working life living standards).<sup>169</sup>

 $^{165}$  Data from the English Longitudinal Study of Ageing WAVE 3, based on data collected in 2006/2007

<sup>167</sup> Calculated using www.minimumincome.org.uk and deducting rent and water rates

<sup>168</sup> Median non-age specific earnings, Labour Force Survey, Oct – Dec 2011, EARN04: Gross weekly earnings of full-time employees (last updated February 2012) www.ons.gov.uk/ons/publications/re-reference-tables.html?edition=tcm%3A77-222482

<sup>169</sup> Pensions Commission (2004) and PPI calculations

<sup>&</sup>lt;sup>166</sup> As defined by Pensioners in focus groups, see www.minimumincomestandard.org for further information on how the Minimum Income Standard is calculated

life living standards by income in working life (2011 earnings terms) <sup>170</sup>	
Income in working life	Target replacement rate
Less than or equal to £11,540	80%
£11,541 - £21,250	70%
£21,251 - £30,360	67%
£30,361 - £60,720	60%
£60,721 or greater	50%

Table 4: Assumed target replacement rates needed to replicate workinglife living standards by income in working life (2011 earnings terms)

Although replacement rates are higher for low income individuals, they may generally find it easier to meet replacement rate targets as a larger proportion of their retirement income is derived from state pensions. Higher earning individuals who do not meet their target replacement rate will not necessarily have incomes that might be considered low. For example, they may still have incomes well above poverty levels.

The analysis only compares state pension and benefits and private pension income against the target income thresholds. In reality, individuals (and in particular higher income individuals) may have other assets, such as savings or housing wealth that they may be able to use to help meet target income levels.

A key assumption of this analysis is that individuals who are aged between 50 and SPA in 2011 and are still working, continue in work, and continue to earn and contribute into their pensions at current levels (allowing for growth in average earnings) until they reach target levels of income.

The analysis also assumes that those who are not currently saving into a pension are, if eligible, auto-enrolled in 2012 and contribute at the minimum required levels. In reality, for various reasons, including health problems, caring responsibilities and lack of available and appropriate employment, not everyone aged 50 and above will be able to continue working and saving until or beyond their SPA.<sup>171</sup>

<sup>&</sup>lt;sup>170</sup> Pensions Commission (2004) and PPI calculations

 $<sup>^{\</sup>rm 171}$  The impact on the results of assuming that all individuals retire two years before SPA are shown in the Modelling Appendix

### Box 3: Uses and limitations of scenario analysis

Scenarios are useful to explore outcomes on the basis of certain assumptions (i.e. to answer questions like: *What would happen if everyone* worked and saved at current levels up until their State Pension Age?). These scenarios are intended as illustrative rather than as forecasts. We do not estimate the likelihood of each scenario occurring.

The projections illustrated in this paper should not be taken as forecasts. The analysis seeks to illustrate the potential impact of working and saving behaviour. However, the decisions that people make and their incomes in retirement will be impacted by personal circumstances (e.g., health, caring responsibilities), employer factors (e.g., the availability of appropriate work), pension provision (e.g., the performance of funds, whether people opt out of pension saving), changes in the structure of the state pension and means-tested benefits system (e.g., impact of the singletier pension and changes to welfare). If a forecast were to be made of levels of income adequacy among future pensioners, it would need to allow for factors that are not allowed for in this paper.

Further details of the methodology and assumptions used in this analysis, and analysis of the sensitivity of results to key assumptions are contained in the Modelling Appendix to this report.

All of the results shown in this chapter have been rounded to the nearest 5%. Totals may not sum to 100% because of this rounding.

**The analysis only covers those in work, or with a partner in work** The following results are based on analysis of the English Longitudinal Study of Ageing (ELSA), so findings apply to England only.

The analysis focuses on individuals aged between 50 and State Pension Age (SPA) who are either in work, or who have a partner who is in work at the time they were interviewed. Of the individuals in the ELSA data aged between 50 and SPA, 80% fall into this category. The remaining 20% of cases are not included in this analysis, as it is unlikely that a majority of these people will enter work again and so working longer is not likely to affect their retirement incomes.

The English population in 2011 aged between 50 and SPA is estimated to be around 7.9 million.<sup>172</sup> The sample used in this project broadly represents 80% of people aged between 50 and SPA, or approximately 6.3 million people.

172 ONS 2010-based subnational population projections

How long will people need to work to meet a basic minimum standard of living in retirement and to replicate working life living standards? Based on the above assumptions the next section of this report considers

how many individuals may be able to meet the Minimum Income Standard from state and private pension savings.

Around 85% of people aged between 50 and State Pension Age in 2011 and still in work could have sufficient state and private pension income to meet the Minimum Income Standard by State Pension Age

- The vast majority, around 85%, (Chart 12) of those aged between 50 and SPA in 2011 and still in work might have sufficient state and private pension income to meet the Minimum Income Standard (calculated to be £211pw for a single pensioner Before Housing Costs (BHC), and £303pw for a couple in 2011)<sup>173</sup> by SPA assuming that they continue to work and save until SPA and that everyone who is entitled to means-tested benefits claims them.
- Around 10% of those aged between 50 and SPA in 2011 and still in work will only be able to meet the Minimum Income Standard if they continue to work and save for a further one to five years after SPA.
- Around 5% of those aged between 50 and SPA in 2011 and still in work might need to work for six years or more after their SPA in order to meet the Minimum Income Standard.

#### Chart 12174



 $^{173}$  Calculated using www.minimumincome.org.uk and deducting rent and water rates  $^{174}$  PPI Dynamic Model – numbers may not sum to 100% due to rounding

This reason so many people are able to meet the Minimum Income Standard is that people who have very low incomes in retirement are generally eligible to receive some support from the state through meanstested benefits. Someone who receives the maximum rate of Guarantee Credit of £138pw, average Housing Benefit and Council Tax Benefit would have an income of £211pw,<sup>175</sup> which is at or just above the Minimum Income Standard. Those who don't have incomes at or above the Minimum Income Standard tend to have assets that exclude them from eligibility for means-tested benefits.

The above analysis assumes that all those eligible for Pension Credit claim it. However, around one million of the four million people in the UK who are entitled to Pension Credit do not claim it<sup>176</sup> and may therefore be living on an income below the Minimum Income Standard level. The following analysis illustrates the difference that claiming Pension Credit can make for the income adequacy of individuals (Chart 13).

Chart 13177



### Means-tested benefits can play an important role in helping those on low incomes during working life achieve minimum acceptable standards of living in retirement

Around 85% of people currently aged between 50 and SPA in 2011 and still in work could meet the Minimum Income Standard by SPA if it is assumed that everyone who is eligible for Pension Credit claims it (Chart

175 www.minimumincome.org.uk/

<sup>176</sup> www.ageuk.org.uk/money-matters/claiming-benefits/pension-credit/

<sup>177</sup> PPI Dynamic Model - numbers may not sum to 100% due to rounding

13). However, if it is assumed that no one claims Pension Credit then only 80% of people aged between 50 and SPA in 2011 and still in work could meet the Minimum Income Standard by SPA (Chart 13).

Claiming Pension Credit allows a further 5% of those aged between 50 and SPA in 2011 and still in work to achieve the Minimum Income Standard by SPA, showing that means-tested benefits can play an important role in helping those on low incomes during working life achieve minimum acceptable standards of living in retirement.

#### Box 4: Adequacy and the Minimum Income Standard

The Minimum Income Standard level will not necessarily be high enough to provide the minimum acceptable standard of living to everyone

- Those with health problems or who live in rural areas<sup>178</sup> may require a higher income to achieve a minimum acceptable standard of living.
- For those who have lived on a higher income during working life the minimum acceptable standard of living is unlikely to *feel* acceptable as it will represent a drop in living standards.
- Many people who receive a level of income equal to or above the Minimum Income Standard may forgo some essentials because of spending in other areas (either discretionary spending on leisure goods or necessary spending, for example, supporting dependents, health related spending or spending on emergencies).

Therefore, just because someone meets the Minimum Income Standard it cannot necessarily be assumed that they are not suffering from deprivation in some capacity.

### It will be harder for people to meet target replacement rates than the Minimum Income Standard

Fewer people will be able to meet a target replacement rate of working life income that would allow them to replicate working life living standards, by their SPA. The exact percentage of working life income needed to meet a target replacement rate varies for each individual depending on their income (Table 4).

If it is assumed that people aged between 50 and SPA in 2011 and still in work, continue to work and save at their current levels, then:

- By SPA, around 40% of those aged between 50 and SPA in 2011 and still in work might have sufficient state and private pension income to meet a target working life replacement rate (Chart 14).
- Around 10% of those aged between 50 and SPA in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further one to five years after SPA.

<sup>&</sup>lt;sup>178</sup> Because of higher transportation costs and less access to discount shops and services

- Around 5% of those aged between 50 and SPA in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further six to ten years after SPA.
- Around 45% of those aged between 50 and SPA in 2011 and still in work, might need to work and save for 11 or more years after their SPA in order to meet a target replacement rate of working life income.



The above analysis shows how close people could come to meeting their target replacement rates through state and private pension income and means-tested benefits. However, some of these people may have other income and assets in retirement which would allow them to meet their target replacement rates when combined with their state and private pension income.

### It will be easier for people to meet the Minimum Income Standard by SPA or during retirement than to meet target replacement rates

Overall it will be easier for people to meet the Minimum Income Standard (£211pw for a single pensioner (BHC); £303pw for a couple in 2011)<sup>180</sup> in retirement through state and private pension income than target replacement rates which are likely to be higher at between 50% to 80% of an individuals working life income (Chart 15).

<sup>&</sup>lt;sup>179</sup> PPI Dynamic Model – numbers may not sum to 100% due to rounding

<sup>&</sup>lt;sup>180</sup> Calculated using www.minimumincome.org.uk and deducting rent and water rates



People with different characteristics may be more or less likely to be able to meet their target replacement rates from state and private pension income by State Pension Age (SPA). The next section of this chapter explores how different characteristics such as income level, gender, and type of pension savings can impact people's likelihood of meeting their target replacement rate by their SPA.

## Those in lower income quartiles may be more able to meet a target replacement rate of working life income by State Pension Age

The following analysis shows how long people from different income quartiles may need to work and save in order to have sufficient state and private pension income to meet a target replacement rate by SPA or after their SPA.

It may be easier for people with very low incomes to meet a target replacement rate of working life income because their target will be much lower than the target of someone who had a high income during working life. For example,

• The income of an individual in the bottom quartile would be equivalent to around £14,000pa or below and their target replacement rate would be 70% of this income at £9,800pa or below (however, as this is lower than the Minimum Income Standard (£11,000pa) this individual would in this analysis have their target replacement rate increased to meet the Minimum Income Standard).

<sup>181</sup> PPI Dynamic Model – numbers may not sum to 100% due to rounding

• The income of an individual in the top quartile would be equivalent to around £35,000pa or above and their target replacement rate would be 60% of this income at £21,000pa or above.

### Chart 16182



Chart 16 shows that:

- Around 70% of those aged between 50 and SPA in 2011 and still in work in the lowest income quartile could have sufficient state and private pension income to meet a target replacement rate by SPA. They are likely to have much lower target replacement rates which they are more able to meet with state pension and means-tested benefits income.
- Only around 30% of people aged between 50 and SPA and still in work in the highest income quartile could meet their target replacement rates (which are likely to be at a much higher income level than those in lower income quartiles) by SPA. In order for those in higher income quartiles to be able to meet their replacement rates, they may need to supplement their state pension (and any state benefits) income and income from private pension with income from other savings and assets, which people in higher income quartiles are more likely to have.

<sup>182</sup> PPI Dynamic Model – income quartiles of benefit units who qualify (by being aged between 50 and SPA in 2011 and still in work) derived from ELSA data using PPI calculations – numbers may not sum to 100% due to rounding

#### Single women and people in couples are more likely to meet a target replacement rate than single men

Chart 17 shows how many people aged between 50 and SPA in 2011 and still in work may be able to meet their individual target replacement rates by SPA and afterwards.

#### Chart 17183



Single women and people in couples are more likely to meet a target replacement rate by SPA than single men.

- Around 50% of single women might have sufficient state and private pension income to reach a target replacement rate of working life income by their SPA, compared with around 35% of single men.
- Around 40% of single women and around 60% of single men might need to work and save for 11 years or more after their SPA to be able to reach a target replacement rate of working life income, compared with around 40% of people in couples.

Women are often earning at lower levels than men and will therefore have lower target replacement rates, making their target replacement rates easier to achieve through state pension, benefit income and minimal pension savings. For example, the median weekly income for a woman in 2011 is around £400pw and for a man in 2011 is around £500pw.<sup>184</sup> The target replacement rate for a median earning woman would be 70% of her

<sup>184</sup> Median non-age specific earnings, Labour Force Survey, Oct - Dec 2011, EARN04: Gross weekly

<sup>&</sup>lt;sup>183</sup> PPI Dynamic Model – numbers may not sum to 100% due to rounding

earnings of full-time employees (last updated February 2012) www.ons.gov.uk/ons/publications/rereference-tables.html?edition=tcm%3A77-222482

income, at around £14,600pa (£280pw) and for a median earning man would be 67% of his income, giving a higher income requirement of around £17,400pa (£335pw). People with very low incomes during working life may be able to meet a target replacement rate through means-tested benefits, and this may disproportionately represent single women.

People in couples may find it easier than single men to meet a replacement rate of working life income as they may be able to supplement their pension income with income from the earnings, savings or pensions of their partner.

• Only around 40% of people in couples may need to work and save for 11 years or more after their SPA to have sufficient state and private pension income to reach a target replacement rate of working life income, compared with around 60% of single men.

However, this is based on the position at the point of retirement, and only considers the joint income while both partners in a couple are alive. Widow(er)s may have lower incomes in retirement after bereavement. These findings are also based on both partners making a joint retirement decision and retiring at the same time.<sup>185</sup>

<sup>185</sup> The impact of assuming that partners in a couple make individual retirement decisions based on individual incomes is shown in the Modelling Appendix

People with Defined Benefit pension savings might find it easier to meet a target replacement rate in retirement

Chart 18 illustrates the proportions of people with different levels and types of pension saving who could meet a target replacement rate by SPA or after.

Chart 18186



People aged between 50 and SPA in 2011 and still in work who have prior Defined Benefit savings (which could also include additional Defined Contribution savings) are much more likely to be able to meet a target replacement rate.

Around 50% of those with prior DB savings could have sufficient state and private pension income to meet their target replacement rate by SPA, while only around 30% of those with prior DC savings (and no DB savings) and only around 25% of those who have no prior pension savings but are auto-enrolled at minimum levels could meet their target replacement rate by SPA.

People aged between 50 and SPA in 2011 and still in work with DB pension savings are less likely to have to work and save for long periods after their SPA in order to meet their target replacement rate. Around 30% of those with DB savings, 60% of those with DC savings (but no DB) and 65% of those with no prior pension savings before auto-enrolment

<sup>&</sup>lt;sup>186</sup> PPI Dynamic Model – numbers may not sum to 100% due to rounding

may need to work and save for 11 years or more after their SPA in order to meet target replacement rates.

Contributing at higher levels to a pension could mean that people with Defined Contribution pensions do not need to work as long to meet their target replacement rates

DB savers have a higher chance of meeting a target replacement rate by SPA partly because they and their employers on average make higher contributions to their pensions.

The following analysis looks at people aged between 50 and 54 and in 2011 and still in work who do not have any DB savings. The analysis assumes that:

- Those who are already saving into DC pensions double their existing contributions; and,
- All auto-enrolled employees and their employers contribute at double the minimum level phased in between 2012 and 2019, (bringing their total contribution level to 16% of band earnings, including tax relief,<sup>187</sup> by 2019) and continue working and contributing until they meet their target replacement rates.

The people in this analysis are aged between 50 and 54 in 2011 and therefore have at least five to ten years to save into a pension (depending on their gender) and some have up to 15 years before their SPA in which to see some benefit from doubling their contributions.

However, this particular cohort of people will not benefit fully from the introduction of auto-enrolment. They will have already have completed most of their working lives before auto-enrolment is introduced, and will also only make and receive low levels of contribution for the first few years after the introduction of auto-enrolment, as the full level of employer, employee and Government contributions will not be in place until 2019. For individuals in this cohort that are auto-enrolled, doubling contributions may have little impact as initial contribution levels are very low.

If analysis was done on people in their 20s and 30s, and it was assumed that all those who were saving in DC pensions paid double contributions, it is likely that the difference in the proportion of people able to meet target replacement by their SPA under the different scenarios would be greater, because they would have several decades in which to accrue a larger pension pot with double contributions. Younger people who are auto-enrolled will also have more years to benefit from saving at the fully phased in contribution levels.



If it is assumed that of those aged between 50 and 54 in 2011 and still in work who are saving in a DC pension or are auto-enrolled (with no DB savings), all DC savers double their contributions and those who are auto-enrolled and their employers contribute at double the minimum level,<sup>189</sup> then 35% of these people might have sufficient state and private pension income to meet their target replacement rate by SPA compared to 30% under minimum contributions (Chart 19).

Under the double contribution scenario around 45% of those aged between 50 and 54 in 2011 and still in work (with no DB savings), may need to work and save for 11 years or more after their SPA to meet their target replacement rate. This is around 15% fewer than under the current and minimum contributions scenario.

<sup>188</sup> PPI Dynamic Model – numbers may not sum to 100% due to rounding

<sup>189</sup> bringing the total contribution level to 16% of band earnings (including tax relief) 8% employee, 6% employer, around 2% in tax relief

### Conclusions

In order to examine how working longer can affect the adequacy of retirement income, the PPI has modelled the retirement income that individuals aged between 50 and State Pension Age (SPA) in 2011 might expect to achieve if they continue to work and save at current levels, up until or beyond their SPA.

This paper uses two measures of income adequacy in order to test whether pensioners might have enough income in order to meet their needs in retirement:

- The **Minimum Income Standard**, which calculates how much income pensioners require to meet the *minimum acceptable standard of living*;<sup>190</sup> *and*,
- Working life replacement rates which calculate how much income an individual pensioner might need in order to achieve a similar standard of living to the one they had in working life.

Meeting the Minimum Income Standard will be easier for people than meeting a target replacement rate of working life income:

- The vast majority, around 85%, of those aged between 50 and SPA in 2011 and still in work might have sufficient state and private pension income to meet the Minimum Income Standard by SPA assuming that everyone who is entitled to means-tested benefits claims them.
- Means-tested benefits can play an important role in helping those on low incomes during working life achieve minimum acceptable standards of living in retirement.
- Around 10% of those aged between 50 and SPA in 2011 and still in work will only be able to meet the Minimum Income Standard if they continue to work and save for a further one to five years after SPA.
- Around 5% of those aged between 50 and SPA in 2011 and still in work might need to work for six years or more after their SPA in order to meet the Minimum Income Standard.

Fewer people will be able to meet a target replacement rate of working life income that would allow them to replicate working life living standards, by their SPA:

- By SPA, around 40% of those aged between 50 and SPA in 2011 and still in work might have sufficient state and private pension income to meet a target working life replacement rate.
- Around 10% of those aged between 50 and SPA in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further one to five years after SPA.
- Around 5% of those aged between 50 and SPA in 2011 and still in work may be able to meet their target replacement rate if they continue to work and save for a further six to ten years after SPA.

<sup>&</sup>lt;sup>190</sup> As defined by pensioners in focus groups

• Around 45% of those aged between 50 and SPA in 2011 and still in work, might need to work and save for 11 or more years after their SPA in order to meet a target replacement rate of working life income.

People with certain characteristics may be more or less likely to be able to meet a target replacement rate by SPA from state and private pension income:

- Those in lower income quartiles may be more able to meet a target replacement rate of working life income by SPA.
- Single women and people in couples are more likely to meet a target replacement rate than single men.
- People with Defined Benefit pension savings might find it easier to meet a target replacement rate in retirement.
- Contributing at higher levels to a pension could mean that people do not need to work as long to meet a replacement rate of working life income which would allow them to replicate working life living standards.

### Chapter four: how might different patterns of work and retirement impact the pension income of individuals?

This chapter analyses how different patterns of working could affect the pre-retirement and post-retirement income of individuals who have different earnings levels.

This chapter investigates the impact of working longer and not working longer on 3 hypothetical individuals who have different earning levels and saving histories during their working life. The three individuals are outlined in boxes 5-7. The working patterns provided in the boxes are a 'baseline' scenario for each individual. The analysis examines the outcomes of different patterns of work, and then compares them to the baseline scenarios to see the impact of different working patterns on people's incomes.

### Box 5: Low earning man - baseline scenario

- He is born in 1963 and begins work at age 22.
- When working, he earns at the 30<sup>th</sup> percentile of age-specific earnings for men.
- While working, he contributes to a DC pension at the average level of 9% of gross earnings (employee and employer contributions).<sup>191</sup>
- He retires in 2030, at his State Pension Age (SPA) of 67 under current proposed policy.<sup>192</sup>

### Box 6: Median earning woman - baseline scenario

- She is born in 1963 and begins work at age 22.
- When working, she earns at median age-specific earnings for women.
- Between the ages of 30 and 35 she takes time out of work to care for her children.
- When she returns to work, she works part-time for two years at 50% of full-time earnings, and then resumes full-time work.
- While working, she contributes to a DC pension at the average level of 9% of gross earnings (employee and employer contributions).
- She retires in 2028, two years prior to her SPA of 67 under current proposed policy and takes her private pension.<sup>193</sup>

<sup>191</sup> Average contributions to a DC occupational pension in 2010, 9% total average contribution made jointly by employer and employee, www.ons.gov.uk/ons/rel/pensions/occupational-pension-scheme-survey-annual-report/2010-annual-report/art-opss2010.html#tab-Chapter-4--Contributions-to-schemes

<sup>192</sup> The Government announced in the Chancellor of the Exchequer's Autumn statement 2011 that they would raise the State Pension Age to 67 between April 2026 and April 2028

<sup>193</sup> In 2010, 35% of women between ages 55-59 had already left the labour market (ONS Pension Trends Ch4 2011). This means that currently many women do not work up until their SPA.

### Box 7: High earning man - baseline scenario

- He is born in 1963 and begins work at age 22.
- When working, he earns at the 70<sup>th</sup> percentile of age-specific earnings for men.
- While working, he contributes to a DC pension at the average level of 9% of gross earnings (employee and employer contributions).
- He retires in 2030, at his SPA of 67 under current proposed policy.

### Box 8: Uses and limitations of hypothetical case study analysis

Hypothetical case studies are useful for looking at how certain individuals may fare under certain assumptions, however these case studies should not be considered predictions of how any particular income group will fare in the future. Each hypothetical individual has a specific history of working and saving behaviour and the behaviour and experiences of any individual in future may be very different from those of the case study individuals.

The assumptions in these case studies are based on the state and private pension and state benefits system as it exists at the time of publishing this paper. The state and private pensions landscape may experience changes in future that are not accounted for in the assumptions in this paper.

### The scenarios in the boxes above represent the baseline scenarios all other alternative scenarios will be compared against

Each individual will be considered under alternative scenarios, under which their working patterns, ages of taking their pension and ages of retirement are varied. The income that individuals receive both before and after State Pension Age (SPA) in each of these alternative scenarios will be compared to the incomes they receive under their baseline scenario.
## **Box 9: Individual modelling assumptions**

- In order to illustrate how levels of income could differ between different scenarios, all individuals are assumed to purchase a single life, level annuity with their private pension saving. While the method people use to take income from pension savings may look very different in future, an annuity provides a good way to measure the income in different scenarios against each other, rather than as an actual projection of income.
- Everyone takes their tax free lump sum and does not use it to provide a secure income in retirement (this is the most common event in practice).
- The current state pension system and the current system of workingage benefits are both in place rather than Universal Credit and/or single-tier state pension.
- The baseline scenarios assume the SPA rises to age 66 by 2020 and rises from age 66 to age 67 between 2026 and 2028 as announced in the Chancellor of the Exchequer's Autumn 2011 statement.

## How might retiring early impact on an individual's income?

The following analysis assumes that the **low earning man** retires at age 60, 7 years before his State Pension Age (SPA) of age 67, due to a health problem which means he is no longer capable of working. The analysis assumes two different scenarios:

- the low earning man purchases a level annuity,
- the low earning man purchases an enhanced annuity on the basis of his health problems. The enhanced annuity offers an uplift of 19%<sup>194</sup> from the rate he would have received from a level annuity.

The low earning man's total gross state and private pension income under these scenarios is compared to the total gross state and private pension income under his baseline scenario in which he retires at his SPA of age 67 in 2030.



# Leaving work before State Pension Age can result in lower income both before and after State Pension Age

When the low earning man is able to remain in work until his SPA of age 67 in 2030 he is able to save into his pension for longer than in the scenarios in which he has to leave work 7 years before SPA due to health reasons. Working and saving for those extra 7 years means his total gross state and private pension income at the point of retirement would be  $\pounds$ 220pw, bringing him within around  $\pounds$ 30 of his target replacement rate of around  $\pounds$ 250pw when he reaches his SPA of age 67 (Chart 20).

Under an alternative scenario, he is assumed to leave work at age 60 on grounds of ill-health. Between age 60 and his SPA of age 67, his income is considerably lower than in the baseline scenario at around £125pw. His weekly income at the point of retirement increases to around £190pw under this scenario once he can claim his state pension, which is around £60 short of his replacement rate.

If instead, the man shops around and purchases an enhanced annuity, his private pension income at the point of retirement would be around £3 higher per week than when he purchases a level annuity, an uplift of 19%, which brings him slightly closer to his replacement rate (Chart 21).

195 PPI Individual Model



# Receiving disability benefits can help those who leave work early to achieve a more acceptable standard of living

The previous analysis showed the low earning man's income without the additional disability benefits he may be entitled to. This is because disability benefits are assumed to be spent on the additional care and support needed by those with disabilities and cannot necessarily be considered as disposable income. For illustrative purposes, Chart 22 shows the low earning man's weekly income with the addition of Disability Living Allowance (DLA) of around £70pw.<sup>197</sup>

<sup>196</sup> PPI Individual Model

<sup>197</sup> DLA care component, £73.60, with no mobility component, This is roughly equivalent to the average DLA award for 60-64 year olds, according to DWP data, February 2011 – DWP tabulation tool, http://83.244.183.180/100pc/tabtool.html



The addition of DLA brings the low earning man's total gross state and private pension income above his target replacement rate for the first 7 years after he reaches SPA and claims his state pension. However, the additional income that he receives through the DLA is paid to meet additional costs, such as care and support. While the income from disability benefits cannot necessarily be considered as disposable income, it is important to consider the effect on his income. If the low earning man did not receive disability benefits he may need to pay for his care and support with his other income thereby reducing his ability to spend on other living costs and reducing his standard of living. Therefore, disability benefits can play an important role in safeguarding against further reductions in income for those who have to leave work early due to health problems.

# Remaining in work until State Pension Age can help to maintain living standards up until retirement

The following analysis assumes that the median earning woman leaves full-time work at age 65, and works part-time until she reaches her SPA of age 67. She takes her private pension at age 65 in order to supplement her income and takes her state pension at age 67. Her income in this scenario is compared to her income in her baseline scenario in which she leaves work at age 65 and takes her private pension (taking her state pension at age 67).

198 PPI Individual Model



In the median earning woman's baseline scenario, she gives up work completely at age 65 and takes her private pension. Her total gross state and private pension income of around £70pw drops below her target replacement rate of around £230pw by around £160pw for two years until she is able to claim her state pension. She may therefore experience a drop in living standards for two years before her SPA, unless she is able to supplement her income from other sources.

In the alternative, phased retirement scenario, in which she works parttime she is able to use her private pension to supplement her earnings in order to maintain a standard of living closer to the one she had in fulltime working life until she reaches her SPA, at which point she is able to claim state pension and use it as a substitute for her earnings (Chart 23).

However, in neither the baseline nor the phased retirement scenario does she remain above her target replacement rate throughout her retirement as, during retirement, her income declines in earnings terms. She may have needed to save for longer in her private pension or contributed a greater proportion of her earnings in order to remain at or above her target replacement rate throughout her retirement. How long might someone need to work in order to replicate working life living standards in retirement?

The following analysis assumes, in the baseline scenario, that the high earning man works full-time until his State Pension Age (SPA) of age 67 in 2030 and then retires completely from work.

In an alternative scenario, the high earning man is assumed to work and save for 2 years after his SPA and take his state and private pensions at age 69.



A high earner, contributing at average levels of salary into a DC pension may need to work beyond State Pension Age in order to meet and sustain his target income during retirement

If the high earner retires at his SPA of age 67 after contributing to his private (DC) pension (with his employer) at average levels for a DC occupational pension (of 9% of his total salary) his total gross state and private pension income will just miss his target replacement rate of around £380pw at retirement, by around £10pw, though the value of his income will continue to reduce in relation to his target replacement rate, in earnings terms, throughout his retirement (Chart 24).

Working full-time and saving for an extra two years, until age 69, allows the high earning man to remain above his target replacement rate for a further 7 years, until age 75, thereby being able to replicate working life living standards well into his retirement. High earners who are

<sup>200</sup> PPI Individual Model

contributing at average levels to their pension may need to work after SPA in order to ensure that they can sustain living standards for longer in retirement.

## Working after State Pension Age can increase net income as a result of not having to pay National Insurance contributions

Part of the reason that people can replicate working life living standards in retirement on lower incomes than they had in working life is because people over SPA are not required to make National Insurance contributions. Therefore, the previous charts have shown illustrations of gross state and private pension income in retirement in comparison to gross target replacement rates. In order to illustrate the way in which not having to pay National Insurance contributions can benefit those who work after SPA, Chart 25 shows the high earning man's net weekly income under a scenario in which he works for two years after his SPA of age 67.

Chart 25201



If the high earning man works for two years after his SPA, his net weekly income is around £190pw higher at his SPA of age 67 than it is under the scenario in which he retires at age 67. He also benefits after he has left work; at the point of retirement (age 69) his net state and private pension income is around £60pw higher than it is under the scenario in which he retires at age 67, as a result of saving for an extra two years into his private pension and deferring his state pension (Chart 25).

# How might rises in State Pension Age affect the individual's retirement income?

The following scenarios explore the potential impact of rises in the State Pension Age (SPA) on the hypothetical individual's income. The scenarios explore potential variations in retirement income for two individuals under the previous legislation in the Pensions Act 2007, under which their SPA would have remained at age 66 in 2030 (rising to 67 between 2034 and 2036), and under the proposed policy which increases their SPA to age 67 by 2028.<sup>202</sup> The aim of these scenarios is to demonstrate how the income of individuals might be affected by the proposed rise in SPA of one year, both if they do or do not work longer to compensate.

#### How might someone who needs to leave work early due to health problems be affected by rises in the State Pension Age?

The following analysis assumes that the low earning man's SPA does not rise to age 67 as under current proposals but instead remains at age 66 in 2030 (not rising to 67 until 2036) as under previous legislation.

The analysis compares the low earning man's income before and after his SPA with the income he receives under the baseline scenario in order to illustrate the potential impact SPA rises may have on the income of someone who is unable to adjust their working patterns to compensate for rises in SPA and cannot work up until their SPA.

<sup>202</sup> The Government announced in the Chancellor of the Exchequer's Autumn statement 2011 that they would raise the State Pension Age to 67 between April 2026 and April 2028



## A rise in State Pension Age could reduce income for those who cannot work longer

Under a scenario in which the low earning man's SPA is one year higher, at age 67, he needs to wait an extra year for his state pension. The rise in SPA does not reduce the income he receives from his private pension because in both scenarios he takes it at age 60 when he is no longer capable of work due to his health problems. However, later eligibility for SPA does mean he needs to live for one year longer on income from his private pension and state benefits, without the benefit of state pension to top his income up and bring him closer to his working life replacement rate.

Therefore, the rise in SPA of one year means that the low earning man must live for a further year on a total gross state and private pension income of around £120pw, which is around £130pw less than his target replacement rate income of £250 per week, until he starts to receive his state pension which brings him to within £70pw of his replacement rate, at age 67 (Chart 26).

# How might a State Pension Age rise affect the income of someone who is able to adjust by working longer?

The following analysis assumes that the **median earning woman** works full-time until age 64. At age 64 she takes her private pension and then works part-time for two years before retiring at her SPA of age 66 and taking her state pension.

Her income under this scenario is compared to her income under a variant scenario in which she leaves full-time work at age 65, and from age 65 takes her private pension and works part-time until she reaches her SPA of age 67.

Chart 27204



# A rise in State Pension Age could result in higher income in retirement for those who can work longer

Under the scenario in which the median earning woman's SPA is one year lower, at age 66, she works for a year less than in the scenario in which her SPA is at age 67. Working for one year less has the effect of reducing her income both at the point of retirement and beyond. Though her income drops below her target replacement rate in earnings terms from the year after retirement in both scenarios, she still has a higher income during the extra year she works before reaching her SPA at age 67 and receives a weekly state and private pension income of around £7pw higher throughout her retirement as a result of working and saving into her private pension for one year longer (Chart 27).

<sup>204</sup> PPI Individual Model

## Conclusions

The modelling analysis explored the impact of working longer and not working longer on 3 hypothetical individuals who have different income levels and saving histories during their working life. The modelling analysis found that:

- Leaving work before State Pension Age (SPA) can result in lower income both before and after SPA.
- Shopping around and purchasing an enhanced annuity could increase income in retirement for people with health problems
- Disability benefits can play an important role in meeting income needs for those who have to leave work early due to health problems.
- Remaining in work until SPA can help to maintain living standards up until retirement
- A high earner, contributing at average levels of salary into a Defined Contribution pension may need to work beyond SPA in order to meet and sustain his target income during retirement.
- Working after SPA can increase net income as a result of tax treatment.

The analysis explored how changes in behaviour in response to a rise in State Pension Age can affect income both before and after retirement:

- As expected, a rise in SPA could reduce income for those who cannot work longer.
- But a rise in SPA could result in higher income both before and after retirement for those who can work longer.

# **Appendix: modelling analysis**

This appendix describes the assumptions and methodology for the modelling in this report. The first part, presented in chapter 3, uses a dynamic model of earnings and pension saving to illustrate what current levels of working and saving might mean at an aggregate level for respondents of the English Longitudinal Study of Ageing (ELSA).

The second part, presented in chapter 4, uses PPI's Individual Model to consider a number of more detailed case studies. Both models were developed with grants from the Nuffield Foundation.

#### **General assumptions**

The following assumptions are used in all modelling work in this report:

- Long-term increases in the retail prices index (RPI) of 3.2%;
- Long-term increases in the consumer prices index (CPI) of 2%;
- Long-term annual earnings growth of 4.7% in nominal terms;
- Short-term economic assumptions for RPI, CPI and annual earnings growth in line with Office for Budget Responsibility projections.<sup>205</sup>
- Expected investment returns of 6% in nominal terms, before charges, corresponding to a mixed equity/bond fund;
- Annual management charges of 1.0% of assets under management for DC schemes;
- When auto-enrolled, individuals and their employers are assumed to contribute at least at the minimum levels required under automatic enrolment legislation (a combined contribution of 8% of band salary)
- Auto-enrolled employees are assumed to enter schemes with NEST charges (0.3% annual management charge, 1.8% contribution charge).

These assumptions are the result of consultation with the PPI's modelling review board, which consists of a number of experts in the field of financial modelling.

Further assumptions, specific to different modelling and scenarios are outlined in later sections.

### Analysis of the English Longitudinal Study of Ageing

The English Longitudinal Study of Ageing (ELSA) is an on-going longitudinal study of a range of social indicators of people aged over 50 in England. Chapter 3 of this report presents aggregated analysis of deterministic projections based on current earnings and savings levels in the ELSA Wave 3 dataset, 2006 made using a dynamic micro-simulation model.

This analysis is based upon data and a set of assumptions regarding future behaviour that may not be fully representative. As a result of this, the results should not be taken as forecasts of future levels of income adequacy.

205 OBR (2012)

#### Method and assumptions

This section provides a brief description of the method and assumptions used in this modelling. The main assumptions are then summarised in Box A1.

The individuals considered in the modelling set out in Chapter 3 work are those in the ELSA Wave 3 dataset, aged between 50 and SPA<sup>206</sup>, belonging to a Benefit Unit<sup>207</sup> that has at least one member still in work. This sample consists of approximately 3000 individuals.

The financial data has been adjusted to make it consistent with 2011 earnings levels, which is taken as the base year for the model. In subsequent years of the projection, individuals are aged and their earnings increased in line with average earnings growth. Pension saving is also projected as outlined in Box A1.

In each year of the projection, potential retirement incomes for each individual are calculated, based upon the value of state pension entitlement, and current and deferred private pension entitlement. These are then converted into Benefit Unit incomes, by matching individuals with their partners, so that Pension Credit, Housing Benefit and Council Tax benefit entitlement can be estimated.

Following this, potential Benefit Unit retirement incomes are compared to either a Minimum Income Standard, or a target replacement rate based on working life incomes. Benefit Units that have potential retirement incomes above this level are then assumed to retire. Couples are assumed to retire at the same time based on a comparison of joint potential retirement income against either the pensioner couple Minimum Income Standard, or a target replacement rate based on joint working life income.

Individuals in Benefit Units that do not reach the required adequacy level are assumed to continue working and saving until they do. Whilst it is Benefit Unit incomes that are tested against the income adequacy level, the results are presented at an individual level.

<sup>206</sup> This refers 65 for men and 60 for women.

 $^{207}$  In this analysis, a benefit unit is considered to consist of either a single adult or a couple living in the same household.

#### Box A1: Baseline dynamic modelling assumptions

- Individuals currently in work are able to continue working and saving until their savings reach the target level.
- Individuals currently contributing to a private pension continue to do so at the same rate that they are currently contributing.
- Individuals not currently contributing to a pension, but who will be eligible for auto-enrolment, begin saving at the minimum contribution rates required under legislation.<sup>208</sup>
- Individuals do not access any private pension savings until the target retirement income level has been achieved.
- Individuals saving in DC schemes are assumed to purchase a single life, level annuity after taking their 25% tax-free lump sum.
- Where appropriate, DB members are assumed to convert 25% of their pension into a lump sum.
- Individuals receive their state pensions at their SPA as legislated in April 2012, and proposed policy.<sup>209</sup>
- The current state pension system is assumed to be in place.
- Target replacement rates are calculated as a percentage of gross working life income as set out by the Pensions Commission modelling. No target replacement rates are assumed to be below the Minimum Income Standard level.
- The Minimum Income Standard used in income testing has both the rent and council tax components removed due to variation in these across the ELSA dataset. For this reason, both Housing Benefit and Council Tax benefit have been removed from Benefit Unit incomes for this analysis.
- State pension entitlement not in payment is not recorded on ELSA. We have estimated entitlement, assuming all individuals receive the full amount of BSP. Average S2P/SERPs entitlement, based on PPI Aggregate Model projections is assigned to individuals by contracted out status.
- Members of a couple are assumed to retire at the same time based upon a comparison of joint income working life income against either the Minimum Income Standard for pensioner couples, or a target replacement rate based on potential joint retirement income.

#### Data

The model makes use of financial variables taken from the ELSA Wave 3 dataset. The most important variables considered for each individual concern:

- Earnings
- Pension contributions
- Current and deferred pension entitlement

 $<sup>^{208}</sup>$  Contributions are phased in between 2012 and 2019 to reach 8% minimum total contributions on band earnings by 2019 - between £5,715 and £38,185 (2010/11 earnings terms)

<sup>&</sup>lt;sup>209</sup> The Government announced in the Chancellor of the Exchequer's Autumn Statement 2011 that they would raise the State Pension Age to age 67 between April 2026 and April 2028.

The required data is not always available. There are two main cases where values have been estimated. The first is that where DC contribution rates are missing, the average has been assigned.

Secondly, ELSA contains no data on the level of deferred pension entitlement. In order to estimate this, broad assumptions on average contribution levels, investment returns and accrual rates were used in conjunction with available data on pension type, dates of scheme membership and current gross salary. Where scheme membership and salary data were unavailable, values were randomly assigned using a 'hot-decking' procedure based on the financial wealth quintile of the individual. This method is similar to that used by the IFS in their paper *Estimating Pension wealth of ELSA Respondents*.<sup>210</sup>

In addition to this, approximately 1% of the individuals included in this modelling have some form of pension entitlement from a former spouse. This has not been included in the analysis.

#### Sensitivity to assumptions

Table A1 demonstrates the change to the target replacement rate results to changes to two of the assumptions made in the analysis:

- Individuals are assumed to convert 100% of their pension into income, rather than taking 25% as a lump sum, which is not used to secure an income in retirement as in the baseline modelling.
- Individual estimated S2P/SERPs entitlement is doubled. As outlined in Box A1, individuals are assigned average S2P/SERPs entitlement based on PPI Aggregate Model projections. This is likely to underestimate entitlement for some high earners, that aren't assumed to be contracted out.
- The assumption that individuals in a couple retire in the same year is relaxed. The alternative to this is that the first person to reach SPA in the couple is able to retire as soon as an income test is satisfied based upon their potential retirement income with their partner's working life income. The second person must then wait until they reach SPA before being income tested and able to retire. This assumption can make it easier for the first member of the couple to reach SPA achieve income adequacy in retirement at that age, but can lead to couples where one member is unable to ever achieve income adequacy in retirement.
- The assumption that individuals can continue to work to SPA. For many this may not be realistic. As an alternative we have modelled each individual retiring 2 years before their SPA. As in this sensitivity retirement is 'forced', only the impact on those able to reach the target replacement rate by SPA is considered.

## Table A1: Results of sensitivity analysis

Figures given in brackets represent change from the baseline results and percentages may not sum due to rounding. Each assumption has been changed in isolation. Figures are rounded to the nearest 5%. \*Represents less than 5%.

		Change from baseline arising from				
Achieve target replacement rate by age	Baseline results	Not taking lump sum	Doubling estimated S2P/SERPs entitlement	Change in treatment of couples	Retiring 2 years before SPA	
By SPA	40%	45% (+5%)	40% (+*%)	50% (+10%)	35% (-5%)	
1-5 years after SPA	10%	10% (+*%)	10% (+*%)	5% (-5%)		
6-10 years after SPA	5%	5% (+*%)	5% (+*%)	5% (-*%)		
11+ years after SPA	45%	35% (-10%)	40% (-*%)	40% (-5%)		

#### Table A2: Range of results from sensitivity analysis

Achieve target replacement rate by age	Range of results
By SPA	40% - 50%
1-5 years after SPA	5% - 10%
6-10 years after SPA	5% - 5%
11+ years after SPA	35% - 45%

### Individual modelling

The PPI's individual model uses individual characteristics and working patterns to project retirement income from private pensions, state pensions and other benefits for hypothetical individuals.<sup>211</sup>

Box A2 outlines the assumptions that apply to all individual modelling work carried out in this project.

#### Box A2: Individual modelling assumptions

- All individuals are assumed to purchase a single life, level annuity with their Defined Contribution private pension savings.
- Everyone takes their tax free lump sum and does not use it to provide a secure income in retirement (this is the most common event in practice)
- The current state pension system and the current system of working-age benefits are both in place rather than Universal Credit and/or single-tier state pension.
- The baseline scenarios assume the SPA rises to age 66 by 2020 as set out in the Pensions Act 2011 and rises from age 66 to age 67 between 2026 and 2028 as announced in the Chancellor of the Exchequer's Autumn 2011 statement.

<sup>&</sup>lt;sup>211</sup> For more information on the Individual Model, see PPI (2003) The Under-pensioned

### Individuals modelled

This research investigates the potential impact of future work and retirement patterns on individuals.

Detailed assumptions have been made about the individuals' working and saving behaviours. These are described in boxes A3-A5, including the assumptions made under any alternative scenarios.

#### Box A3: Low earning man

- He is born in 1963 and begins work at age 22.
- When working, he earns at the 30<sup>th</sup> percentile of age-specific earnings for men.
- In 2003, aged 40 he and his employer begin contributing into a DC scheme at the average rate of 9% of gross pay (combined).
- Under the baseline scenario, he retires in 2030, at his SPA of 67 under current proposed policy.<sup>212</sup>
- He is then considered under two alternative scenarios:
  - He is forced into early retirement, aged 60 due to disability. He takes his private pension at this age, also receiving Employment and Support Allowance until he reaches SPA of 67.
  - He is forced into early retirement, as in the first alternative scenario, but the planned rise in SPA from 66 to 67 between 2026 and 2028 does not take place, so he reaches his SPA of 66 in 2029.

<sup>212</sup> The Government announced in the Chancellor of the Exchequer's Autumn statement 2011 that they would raise the State Pension Age to 67 between April 2026 and April 2028

#### Box A4: Median earning woman

- She is born in 1963 and begins work at age 22.
- When working, she earns at median age-specific earnings for women.
- Between the ages of 30 and 35 she takes time out of work to care for her children.
- When she returns to work, she works part-time for two years at 50% of full-time earnings, and then resumes full-time work.
- While working, she contributes to a DC pension at the average level of 9% of gross pay (employee and employer contributions).<sup>213</sup>
- Under the baseline scenario, she retires in 2028, two years prior to her SPA of 67 under current proposed policy and takes her private pension.<sup>214</sup>
- She is then considered under two alternative scenarios:
  - After ceasing full-time work at age 65, she works two years parttime at 50% of full-time earnings in order to supplement her private pension income. She stops working at her SPA of 67.
  - The proposed rise in SPA from 66 to 67 does not take place. Under this scenario, the Median Earning Woman stops full time work aged 64, rather than 65. She takes her private pension at this age and works part-time at 50% of full-time earnings for two years until she reaches her SPA of 66.

#### Box A5: High earning man

- He is born in 1963 and begins work at age 22.
- When working, he earns at the 70<sup>th</sup> percentile of age-specific earnings for men.
- When in work, he and his employer contribute to a DC pension at the average level of 9% of gross pay (employee and employer contributions).
- Under the baseline scenario, he retires in 2030, at his SPA of 67 under current proposed policy.
- Under an alternative scenario he continues to work until age 69, two years after his SPA. He defers his state pensions and delays taking his private pension until retirement at this age.

<sup>&</sup>lt;sup>213</sup> Average contributions to a DC occupational pension in 2010, 9% total average contribution made jointly by employer and employee, www.ons.gov.uk/ons/rel/pensions/occupational-pension-scheme-survey-annual-report/2010-annual-report/art-opss2010.html#tab-Chapter-4--Contributions-to-schemes

 $<sup>^{214}</sup>$  In 2010, 35% of women between ages 55-59 had already left the labour market (ONS Pension Trends Ch4 2011). This means that currently many women do not work up until their SPA.

# Acknowledgements and Contact Details

The Pensions Policy Institute is grateful for input from many people in support of this paper, in particular:

John Adams	
Stephen Balchin	
Nicholas Barr	
Jim Boyd	
Christopher Brooks	
Thomas Q. Christensen	

Niki Cleal Rowena Crawford Maritha Lightbourne Katy Litt Stephen McNair John N. Migliaccio Ian Owen Susan St. John Paul Shelley Sandra Timmermann David Yeandle OBE

Editing decisions remained with the authors who take responsibility for any remaining errors.

The Pensions Policy Institute is an educational charity promoting the study of retirement income provision through research, analysis, discussion and publication. The PPI takes an independent view across the entire pensions system.

The PPI is funded by donations, grants and benefits-in-kind from a range of organisations, as well as being commissioned for research projects. To learn more about the PPI, see: <u>www.pensionspolicyinstitute.org.uk</u>

© Pensions Policy Institute, 2012

Contact: Niki Cleal, Director Telephone: 020 7848 3744 Email: <u>info@pensionspolicyinstitute.org.uk</u>

Pensions Policy Institute King's College 26 Drury Lane London WC2B 5RL

The PPI is grateful for the continuing support of its Platinum and Gold members:

MetLife The Pensions Regulator Prudential UK & Europe Threadneedle Investments AXA Investment Managers BlackRock Capita Hartshead Department for Work and Pensions Hymans Robertson RPMI

A full list of supporting members is on the PPI's website.

This paper uses data from the English Longitudinal Study of Ageing (ELSA). The data were made available through the UK Data Archive (UKDA). ELSA was developed by a team of researchers based at the National Centre for Social Research, University College London and the Institute for Fiscal Studies. The data were collected by the National Centre for Social Research. The funding is provided by the National Institute of Aging in the United States, and a consortium of UK government departments co-ordinated by the Office for National Statistics. The developers and funders of ELSA and the Archive do not bear any responsibility for the analyses or interpretations presented here

The Nuffield Foundation is an endowed charitable trust that aims to improve social well-being in the widest sense. It funds research and innovation in education and social policy and also works to build capacity in education, science and social science research. The Nuffield Foundation has funded the development of the PPI models.

# **References**

Age UK (2012) Flexible employment: how employment and the use of flexibility policies through the life course can affect later life occupation and financial outcomes Age UK www.ageuk.org.uk

Age UK (2010) Consultation response- Call for evidence: Review of the Default Retirement Age Age UK www.ageuk.org.uk

Association of Independent Financial Advisers (AIFA) (2010) *Financial planning through retirement benchmarking study* AIFA

Anderson, P. Gustman, A. Steinmeier, T. L. (1997) *Trends in Male Labor Force Participation and Retirement: Some Evidence on the Role of Pensions and Social Security in the 1970s and 1980s* NBER Working Paper Series, Vol. w6208, 1997. http://ssrn.com/abstract=225966

Banks, J. Emmerson, C. Tetlow, G. (2005) *Estimating Pension Wealth of ELSA Respondents* (IFS) www.ifs.org.uk

'Banks, J. Tetlow, G. (2008) Extending working lives' in Banks, J. Breeze, E. Lessof, C. Nazroo, J. (2008) *Living in the 21st century: older people in England: the 2006 English Longitudinal Study of Ageing, WAVE 3* www.ifs.org.uk/elsa/index.php

Barnes, H. Sissons, P. Stevens, H. (2011) *Employment and Support Allowance: Findings from a follow-up survey with customers* (DWP) www.dwp.gov.uk

Barnes, H. Sissons, P. Stevens, H. (2010) *Employment and Support Allowance: Findings from a face-to-face survey of customers* (DWP) www.dwp.gov.uk

Barrell, R. Kirby, S. Orazgani, A. (2011) *The macroeconomic impact from extending working lives* (DWP) www.dwp.gov.uk

Buckner, L. Yeandle, S. (2006) Who Cares Wins: The Social and Business Benefits of Supporting Working Carers Carers UK www.carersuk.org

Campos, C. Dent, A. Fry, R. Reid, A. (2011) *Regional Trends* 43 – *Impact of the recession* Office for National Statistics (ONS) www.ons.gov.uk

Citizens Advice Bureau (CAB) (2010) Not working: CAB evidence on the ESA work capability assessment CAB www.citizensadvice.org.uk

'Crawford, R. Tetlow, G. (2010) Employment, retirement and pensions' in Banks, J. Lessof, C. Nazroo, J. Rogers, N. Stafford, M. Steptoe, A, Marmot, M. (2010) *Financial circumstances, health and well-being of the older population in England: the 2008 English Longitudinal Study of Ageing, WAVE 4* www.ifs.org.uk/elsa/index.php Cushing-Daniels, B. Steuerle, C. E. (2009) *Retirement and Social Security: A Time Series Approach* Boston College Centre for Retirement Research. http://crr.bc.edu/

Department for Work and Pensions (DWP) (2012a) Employment and Support Allowance – Incapacity Benefits Reassessments: Outcomes of Work Capability Assessments, Great Britain (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2012b) *Employment and Support Allowance: Outcomes of Work Capability Assessments, Great Britain – new claims* (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2011a) A state pension for the 21<sup>st</sup> century (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2011b) *Pension Bill 2011 fact sheet 1* – *Government amendment to Clause 1 of Pensions Bill 2011* (DWP) www.dwp.gov.uk/docs/pensions-bill-spa-info.pdf

Department for Work and Pensions (DWP) (2011c) *The Pensioners' incomes series* 2009-10 (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2010a) *A sustainable State Pension: When the State Pension Age will increase to 66* (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2010b) *Pensions and later life: Communications tracking research October* 2010 (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2010c) A comparative review of international approaches to mandatory retirement (DWP) www.dwp.gov.uk

Department for Work and Pensions (DWP) (2008) *Impact Assessment of The Employment and Support Allowance Regulations 2008* (DWP) dwp.gov.uk/docs/impactassessment180308.pdf

Department for Work and Pensions (DWP) (2006) *The gender impact of pension reform* (DWP) www.dwp.gov.uk/docs/genderimpactassessment. pdf

Dixon, S. (2008) *Transitions from Work to Retirement* Statistics New Zealand and Department of Labour. www.stats.co.nz

Gustman, A. Steinmeier, T. (2007) *Projecting Behavioral Responses to the Next Generation of Retirement Policies* NBER Working Papers 12958, National Bureau of Economic Research

Hedges, A. Sykes, W. (DWP) (2009) *Extending working life: changing the culture Qualitative research into effective messages* (DWP) www.dwp.gov.uk

Hirsch, D. (2005) *Sustaining Working Lives: A Framework for Policy and Practice,* Joseph Rowntree Foundation.

Lain, D. (2011) *Helping the Poorest Help Themselves? Encouraging Employment Past 65 in England and the USA* Journal of Social Policy 40:3, 493–512. Cambridge University Press 2011

Liebman, J. Luttmer, E. (2010) *Would people behave differently if they understood social security? Evidence from a field experiment* National Bureau of Economic Research www.nber.org/papers/w17287

Dr Macleod, A. Worman, D. OBE, Wilton, P. Woodman, P. Hutchings, P. (2010) *Managing an ageing workforce: How employers are adapting to an older labour market* The Chartered Institute of Personnel and Development (CIPD) Chartered Management Institute (CMI)

Mai Sims, J. Vanderaa, K. (2008) European Network Against Racism (ENAR) ENAR shadow report 2008, Racism in the United Kingdom (ENAR) www.runnymedetrust.org/uploads/FINAL%20SR%202008.pdf

Maltby, T (2011) *Extending Working Lives? Employability, Work Ability and Better Quality Working Lives* Social Policy and Society 10:3, 299-308. Cambridge University Press

McNair, S. Flynn, M. Dutton, N. (2007) *Employer responses to an ageing workforce: a qualitative study*. DWP Research Report 455. (DWP) www.dwp.gov.uk

Metcalf, H. Meadows, P. (2010) Second survey of employers' policies, practices and preferences relating to age, 2010 (BIS) (DWP) www.dwp.gov.uk

Metcalf, H. Meadows, P. (2006) *Survey of employers' policies, practices and preferences relating to age, 2010* (BIS) (DWP) www.dwp.gov.uk

New Zealand Ministry of Social Development (MSD) (2009) *To Work, or Not to Work? Findings from a Survey of 65-year-old New Zealanders* (MSD) www.msd.govt.nz

New Zealand Department of Labour (2007) *Older people in work: Key Trends and Patterns* 1991-2005 www.retirement.org.nz/webfm\_send/346

Organisation for Economic Co-operation and Development (OECD) (2011) *Pensions at a Glance 2011* (OECD)

Office for National Statistics (ONS) (2012) *Pension Trends - Chapter 4: The labour market and retirement* www.ons.gov.uk

Office for National Statistics (ONS) (2011a) *Occupational Pension Schemes Annual Report* 2010 www.ons.gov.uk

Office for National Statistics (ONS) (2011b) *Pension Trends - Chapter 7: Pension scheme membership* www.ons.gov.uk

Office for National Statistics (ONS) (2010) *Survey of Carers in Households* 2009/10 www.ons.gov.uk

Office for National Statistics (ONS) (2009) *Impact of the recession on the labour market* www.ons.gov.uk

Phillipson, C. Smith, A. (2005) *Extending working life: a review of the research literature* DWP Research Report 299 (DWP) www.dwp.gov.uk

Smith, S. (2004) *Can the retirement consumption puzzle be resolved? evidence from UK panel data* Institute for Fiscal Studies (IFS) www.ifs.org.uk

Sykes, W. Coleman, N. Groom, C. (2010) *Review of the Default Retirement Age: Summary of the stakeholder evidence* (BIS) (DWP) www.dwp.gov.uk

Turner, J. A. (2007) *Promoting Work: Implications of Raising Social Security's Early Retirement Age* Work Opportunities for Older Americans, Series 12, August 2007. Boston College Centre for Retirement Research. http://crr.bc.edu/

Vickerstaff, S. Loretto, W. Milne, A. Alden, E. Billings, J. White, P. (2009) *Employment support for carers* DWP Research Report 597 (DWP) www.dwp.gov.uk

Vickerstaff, S. (2010) Older Workers: the 'Unavoidable Obligation' of Extending Our Working Lives? Sociology Compass 4:10, 869-879. Blackwell Publishing ltd.

Wachter, T. (2009) *The End of Mandatory Retirement in the US: Effects on Retirement and Implicit Contracts* Columbia University www.columbia.edu

Crown copyright material is reproduced with the permission of the Controller of HMSO and the Queen's Printer for Scotland.



www.pensionspolicyinstitute.org.uk ISBN 978-1-906284-22-0